#### **TABLE OF CONTENTS**

	and Acronyms	
	I on Lesotho's Vision 2020	
Executive Su	mmary	X
	ND DATIONALE	
PURPOSE A	<b>ND RATIONALE</b> troduction	1
	e Process	
1.2.1	Political Commitment	
1.2.1	Consultations	ا
1.2.3	The Preparation of the Document	
	ethodology	
1.4 Th	e Structure of the Document	<u>2</u>
	SION 2020	
	roduction	
2.2 Th	e Lesotho Vision 2020 statement	4
2.3 De	escription of the Vision	4
2.3.1	A Stable Democracy	4
2.3.2	A United Nation	5
2.3.3	A Nation at Peace with Itself and Its Neighbours	5
2.3.4	A Healthy and Well Developed Human Resource Base	5
2.3.5	A Strong Economy and Prosperous Nation	6
2.3.6	A Well Managed Environment	7
2.3.7	A Well Established Technology	7
	nclusion	
	DPMENT ENVIRONMENT	
	roduction	
	sessment of the Development Performance	9
3.2.1	A Stable Democracy	10
3.2.2	A United Nation	
3.2.3	A Nation at Peace with itself and its neighbours	12
3.2.4	A Healthy and Well Developed Human Resource Base	13
3.2.5 3.2.6	A Strong Economy and Prosperous Nation A Well Managed Environment	14
3.2.6 3.2.7	A Well Established Technology	19
	rengths, Weaknesses, Opportunities and Threats (SWOT)	20
3.3 Sti 3.4 Su	mmary of the Challenges	21
CHAPTER 4.		21
	ANALYSIS	
	roduction	
	ur Scenarios	
	e Preferred Scenario	
	nclusion	
NATIONAL D	EVELOPMENT STRATEGY	29
	roduction	29
5.1.1	The Grand Strategy	
5.1.2	Thematic Strategies	30

CHAPTER	R 6	43
IMPLEME	NTATION, MONITORING AND EVALUATION	43
6.1	Introduction	43
6.2	Implementation, Monitoring and Evaluation	
6.3.1		43
6.3.2	Monitoring and Evaluation	44
6.3.3		
6.3	Vision Supporting Institutions	
6.3.1	Functions of Vision Supporting Institutions	48
6.3.2	Implementation, Monitoring and Evaluation Plan -	49
Annexes		
	esotho Vision 2020: Critical Success Factors	
Annex 2: S	Scenario Drivers and Assumptions	52
Annex 3: S	Summary of SWOT	53

#### ABBREVIATIONS AND ACRONYMS

ACP African, Caribbean and Pacific
AGOA African Growth and Opportunity Act
AIDS Acquired Immune Deficiency Syndrome

AU African Union

CBOs Community Based Organisations

CEDAW Convention on the Elimination of All forms of Discrimination Against Women

CET Common External Tariff
CFC Chloro-Flouro Carbon
CMA Common Monetary Area

COSC Cambridge Overseas School Certificate
CHAL Christian Health Association of Lesotho

DDPR Directorate of Dispute Prevention and Resolution

DRT District Resource Teachers

ECCD Early Childhood Care and Development

EFA Education for All

EIA Environmental Impact Assessment
EIP Expanded Immunization Programme

EU European Union

FDI Foreign Direct Investment FPE Free Primary Education

FSSP Food Self-Sufficiency Programme
GATT General Agreement on Tariffs and Trade

GDP Gross Domestic Product
GEF Global Environmental Facility
GNI Gross National Income
GNP Gross National Product
GOL Government of Lesotho
HS Harmonized System
HIV Human Immuno Virus

ICT Information Communication Technology
IEC Independent Electoral Commission
IFC International Finance Corporation

ISP Internet Service Provider

JBCC Joint Bilateral Commission of Cooperation

JC Junior Certificate

LAPCA Lesotho AIDS Programme Coordinating Authority
LCCI Lesotho Chamber of Commerce and Industry

LCN Lesotho Council of NGOs LDF Lesotho Defence Force LDCs Least Developed Countries

LeaRN Leadership Regional Network for Southern Africa

LHWP Lesotho Highlands Water project LMPS Lesotho Mounted Police Service

LNDC Lesotho National Development Corporation
LNOC Lesotho National Olympic Committee

LRA Lesotho Revenue Authority

LSRC Lesotho Sports and Recreation Commission
LTA Lesotho Telecommunications Authority
LTDC Lesotho Tourism Development Corporation

MDGs Millenium Development Goals

MEAs Multilateral Environmental Agreements

M&E Monitoring and Evaluation
MFA Multi Fibre Agreement
MFN Most Favoured Nation

MOHSW Ministry of Health and Social Welfare

MOV Means of Verification

MTEF Medium Term Expenditure Framework

NAC National AIDS Committee

NEPAD New Partnership for Africa's Development NEAP National Environmental Action Programme

NER National Enrolment Ratio

NES National Environment Secretariat

NFE Non-Formal Education

NGO Non-Governmental Organization

NPA National Plan of Action
NSC National Steering Committee
NSS National Security Service
ODA Official Development Assistance
OVI Objectively Verifiable Indicators

PHC Public Health Care

Popn Population

PSIRP Public Sector Improvement and Reform Programme

PRSP Poverty Reduction Strategy Paper

PS Principal Secretary
RSA Republic of South Africa

SADC Southern African Development Community

SAP Structural Adjustment Programme
SARS Severe Acute Respiratory Syndrome

S&T Science and Technology

SACU Southern African Customs Union SMME Small, Micro and Medium Enterprises

SE Secondary Education

SWOT Strengths, Weaknesses, Opportunities and Threats TVET Technical and Vocational Education and Training UNCBD United Nations Convention on Biological Diversity

UNDP United Nations Development Programme

UNFCCC United Nations Framework Convention on Climate Change UNCCD United Nations Convention to Combat Desertification

UNICEF United Nations Children's Fund

USA United States of America
USD United States Dollars
VAT Value Added Tax

WASA Water and Sewage Authority

WEHAB Water, Energy, Health, Agriculture and Biodiversity

WHO World Health Organization WTO World Trade Organization

#### A CLARION CALL ON LESOTHO'S VISION 2020

Four years ago, I beseeched you, my people, to join hands with your government to produce a vision that would guide the development of our country in the years to come. At that time, little did I discern the apposite nature of the request and little did I anticipate the overwhelming enthusiasm that it would receive. This visioning process has unleashed among us a new outlook on life, an increased vigour and resolve, and indeed, a new sense of hope for a better future. We have, all of us, put together a document that aptly captures our dreams and aspirations, and maps out the way forward in the improvement of our lives and that of our children. For this, we deserve to be congratulated.

The completion of this Vision 2020 document is, needless to mention, not the end of our development journey. It is rather a pathway, indeed a roadmap ushering our destiny into our own hands. There are genuine fears that we could remain with our hands clutched witlessly on to this splendid document, with little, if anything at all, by way of implementation. Yet conversely, and judged from all angles, our history is a story of resilience, steadfastness and untold resourcefulness. My personal conviction is that, without any speck of doubt, Basotho have what it takes to deliver on this vision.

I call upon every one of you to give your most fervent and unrelenting support to the implementation of the great ideas that we have laid out in this document. Let Vision 2020 be a band which binds us together and which strengthens our oneness, a catalyst that spurs us on to higher levels of performance and achievement. Indeed, let Lesotho be a stable democracy, a united and a prosperous nation at peace with itself and its neighbours. Let Lesotho pride itself with a healthy and a well developed human resource base, a strong economy, a well managed environment and a technology that is well established. My people, let Lesotho, our beloved Kingdom in the Sky, seize this opportunity and reach for the pinnacle of its development in the coming two decades.

LETSIE III KING OF LESOTHO

#### **FOREWORD**

Our history shows that towards the end of the 18<sup>th</sup> Century, Lesotho was a country of stunning beauty. Its towering mountains and rugged hills, the undulating plains and gorgeous valleys, punctuated by an elaborate system of sinuous rivers and streams, and richly endowed with a variety of indigenous grasses, all combined to make it a land of extreme beauty and splendour. On this land prowled a unique assortment of animals, and a variety of birds, big and small, grazed its skies. Indeed, Lesotho was, in those days, a country of unparalleled elegance and striking beauty.

Although at that time agriculture was already the backbone of Lesotho's economy, the discovery of diamond at Kimberly created an unprecedented demand for grain, especially wheat, as people came from all over Southern Africa to find employment on the mine. Basotho farmers responded with impressive verve to this increase in demand. They switched quickly from the production of sorghum to wheat, and made commendable adjustments in their modes of transport, from draft animals to large carts drawn by cattle. Lesotho quickly became known as the *Granary of Southern Africa*. At this time there was no question of food insecurity.

This great commercial prowess made Lesotho very attractive. Not only did many people in the region migrate to the Kingdom, but its educational institutions also began to attract multitudes of learners from all over the sub-continent and beyond. Lesotho soon became one of the leading countries in literacy on the African continent.

Over the years, however, the situation has deteriorated phenomenally all round. Long years of drought and land degradation have combined with negative political and economic factors to reverse many of the gains described above. Agricultural production has declined to a point where the country now has a serious problem of food security. Wildlife is fast disappearing from our landscape. Poor land use management as well as climate change and a general degradation of the environment have led to aridity and barrenness of the land with an unmistakable signs of the onset of desertification.

This unprecedented deterioration in the land and its people have ushered in a crisis in the social and moral fabric of our society. Unemployment is high, and poverty is deepening; stock theft and armed robbery have become the order of the day; corruption and nepotism are spreading like a cancer in our institutions; and there is no doubt that life is becoming harder by the day for an increasing number of our people. It is a desperate situation that calls for desperate measures.

The Vision 2020 could, therefore, not have come at a more appropriate time. Our people need a rallying point. They need a unifying dream that will galvanize their energies and harmonize their efforts. They need a long-term vision that will help to bring consensus in the allocation of our meagre resources. This document, itself a result of wide consultations among our people, should no doubt serve as a beacon of hope in all our communities. It should unite us and give us a collective focus.

I should point out, however, that this document by itself will not feed us; it will not create prosperity; it will not educate our children; it will not preserve our environment; nor will it see to the advancement of our technology. It is only a planning tool that is intended to guide our development efforts. We need to go out and implement it. It will only be of benefit to us if we make use of it. It should form part of the way we live. It should be an active document that is discussed in many fora and debated during the formulation of strategic plans. It should be kept alive and well.

My fellow citizens, you have heard His Majesty's Clarion Call. Let us now take our destiny into our own hands and move this country into the prosperous future of our dreams.

PAKALITHA B. MOSISILI, MP PRIME MINISTER

#### **PREFACE**

Our Vision 2020 Document presents a broad perspective framework of how Basotho, through their representatives would like to see their country by the year 2020. This represents a fundamental shift from the way we have approached development planning in the past. Starting from 1970, Lesotho based its development on Five Year Development Plans, Annual Plans and Budgets. In 1997 it changed to Three Year Rolling Plans. While these plans were useful in so far as they went, they were not long enough to see where the nation would be in fifteen or twenty years. Such a perspective plan is necessary because development of natural resources such as water, mining, forestry, etc, require more than a five year lead-time to realise full benefits. The second consideration was that the measurement of progress on poverty reduction and employment generation requires a perspective that is longer than three year plans. Finally the Vision provides a beacon towards which national energies and resources can be mobilised. It also facilitates ownership of the development agenda on a sustained basis.

This document thus represents a major improvement over these inadequacies of the past. It is not only all encompassing and sufficiently long-term in nature, it is also a product of discussions with Basotho from all walks of life. Consultations were done at the village, community, district and central levels. Focus groups included political leaders, chiefs, Members of Parliament, district councils, youth groups, people with disabilities, public servants and students. The Steering Committee that compiled it was itself made up of representatives from all sectors of our society. In a nutshell, this document truly reflects the wishes and aspirations of Basotho. Most important of all, the nation came together in the dialogue of June 2004, and unanimously adopted this document as the framework within which all future national plans will be placed.

Lesotho's Vision 2020 document identifies seven pillars of development. These are democracy, unity, peace, education and training, economic growth, management of the environment, and advancement in technology. These pillars are key components of the vision statement. Based on an assessment of past development trends, and an analysis of the pertinent strengths, weaknesses, opportunities and threats, the document goes on to identify factors that are pivotal in Lesotho's development. These are political commitment, foreign direct investment, and public sector management. These factors are then used to develop four possible scenarios; namely, *melupe, khomo-ha-li-kae-batho-re-ba-ngata, metse-e-metle-liotloana, and lesupi.* 

Having identified the vision, together with the different possible scenarios, the document suggests two groups of strategies to achieve the vision. The first group comprises grand strategies that target the three key factors of political commitment, foreign direct investment, and public sector management; while the second group is made up of thematic strategies that target the seven pillars mentioned above. The document concludes by identifying concrete actions that need to be carried out, the key players and mechanisms by which the whole process will be monitored.

As the Minister of Finance and Development Planning, I will ensure that all national development plans and programmes will be based on this Vision. I will also ensure that periodic reviews of progress towards the realisation of the Vision are undertaken.

The King's Clarion Call is loud and clear. The commitment of the Prime Minister is firm and unequivocal. It is now for us all to rise above our differences and undertake this noble development crusade.

### TIMOTHY THAHANE MINISTER OF FINANCE AND DEVELOPMENT PLANNING

#### **EXECUTIVE SUMMARY**

In 2000 the country took a policy decision to formulate a vision to provide a long-term perspective within which national short to medium-term plans could be formulated. The **specific objectives** of the Lesotho Vision 2020 are to: establish a long-term vision for Lesotho by looking beyond the short-term plans and adjustments; explore the options for economic, political and human development to the year 2020; identify alternative development strategies suitable for the Lesotho situation; promote a process of open dialogue and consultation with socio-economic groups countrywide; create an environment whereby Basotho will actively participate in achieving the Vision 2020; and, develop a focus along the horizon in the direction of which development plans could be rolled out.

The **process** of developing the Lesotho Vision 2020 went through three stages, namely; the inception stage for political commitment and support; the nation-wide consultations stage and the formulation stage.

The **methodology** used to develop the Lesotho Vision 2020 is a combination of participatory futures studies, strategic and development planning. The Lesotho Vision 2020 Document addresses four basic questions, which are: how do the Basotho people envision their future? What are the challenges that face the development process? What could happen to the development process in the future? What should be done to realize the Vision?

The document is structured in six chapters. Chapter 1 is an introduction featuring the process and the methodology towards the formulation of Lesotho Vision 2020.

In Chapter 2, the Vision statement is presented and it reads: By the year 2020 Lesotho shall be a stable democracy, a united and prosperous nation at peace with itself and its neighbours. It shall have a healthy and well-developed human resource base. Its economy will be strong; its environment well managed and its technology well established.

The conceptualisation of the Vision is in the same Chapter, driven through a description of its components; namely, a stable democracy, a united nation, a nation at peace with itself and its neighbours, a healthy and well-developed human resource base, a strong economy and prosperous nation, a well- managed environment and a well-established technology.

Chapter 3 on the **development environment** reviews trends in the development performance of Lesotho with the purpose to identify key trends, events and challenges in different sectors of the economy. In addition, Chapter 3 reviews key aspects of the environment as they affect national development prospects. Essentially, the Chapter is an analysis of Lesotho's development environment; and to this end uses as a

framework, the components of the Lesotho Vision 2020 Statement as adopted from the First National Dialogue<sup>1</sup>.

Good governance, media freedom and gender sensitivity should characterize Lesotho as a stable democracy. With regard to governance, Lesotho is a democratic country with the Monarch as head of state, the Prime Minister as head of government and local governance structures. Governance in Lesotho has been strengthened by the recently adopted system that has for the first time in the history of the country enabled ten political parties to be represented in parliament.

The challenges that the Kingdom still has to contend with are: empowering the legislature to carry out the oversight function, improving the judiciary to deal with the delays in the justice system and strengthening the public service for effective service delivery and accountability.

Development management capacity is another aspect of governance. For this, Lesotho has embarked on the Public Sector Improvement and Reform Programme (PSIRP). The programme represents the Government's framework for public sector improvement and reform.

The key challenges in development management capacity include: improving research capacity, coordinating information management systems, dealing with brain drain in different sectors, implementing the PSIRP and strengthening the public-private and civil society partnership in development.

Local governance and popular participation contribute towards good governance. To this end the country is working towards decentralisation by implementing the Local Government Act of 1997.

The challenge is to empower the imminent local government authorities, and to improve chieftainship as a strategy to complement local governance at the grassroots level.

Freedom and pluralism of the Media are some of the measures of stability and democracy. According to the United Nations Reports, media in Lesotho is free from direct government interference.

The challenges here include the consolidation of the currently scattered pieces of legislation regulating the media, improving media independence and professionalism as well as extending media coverage to the rural areas of the country.

- Nation at peace with itself and its neighbours = (peace)

- Well established technology = (information, communication, science, technology)

<sup>&</sup>lt;sup>1</sup> The components of the Vision Statement have been clustered as follows:

<sup>-</sup> Stable democracy = (governance, media, gender)

United nation = (culture)

Healthy and well developed human resource base = (health, HIV and AIDS, education, sports)

Strong economy and prosperous nation = (economy, prosperity)

Well managed environment = (environment)

**With regard to** *gender,* Lesotho's female population enjoys higher education attainment and literacy than the male population. This enables upward mobility for them in various sectors of the economy. The key challenge is to uproot discrimination as a way of life and appoint more women into areas of responsibility in both the public and private sectors without neglecting boys and men.

Lesotho is envisaged to be a **united nation** by 2020. The role of the Monarchy is to unify the nation. The widespread use of national symbols such as the national flag signifies a united nation sharing common goals based on a common cultural heritage. The greatest challenge is to inculcate the spirit of patriotism in all Basotho. Other challenges include protection of Lesotho's culture and its heritage.

Lesotho is a **nation at peace with itself and its neighbours.** Evidence of internal peacefulness of Lesotho lies in: the known peace-loving nature of Basotho, the fact that Basotho are a nation with Sesotho as the main language and a shared history, culture and traditions. Lesotho is also at peace with her SADC and African neighbours. However, the country still faces the challenge to promote political tolerance, and combat the current rate of internal and cross-border stock theft, crime and armed robberies all of which deprive communities of the necessary state of peace.

The Government's efforts to address good health, the HIV and AIDS pandemic and education are indicative of a commitment to the attainment of **a healthy and well-developed human resource base** by 2020. With regard to **health**, Lesotho has made remarkable progress in providing access to social infrastructure, including access to safe drinking water. Challenges include equitable distribution of health facilities and strengthening the health institutions for efficient and effective service delivery.

The Government of Lesotho recognises that *HIV and AIDS* is not only a health problem but a multi-sectoral development issue that has social, economic and cultural implications. Combating further spread of HIV and AIDS therefore continues to be one of the biggest challenges that face the country.

**Education** is central to national development. Presently Lesotho's adult literacy rate (82%, 2002) is higher than in most African countries. The country is committed to provision of an equitable basic education to all Basotho as a key development goal. Key challenges include: further improving access to education at all levels, and developing a curriculum that responds to the national development priorities, thus promoting entrepreneurial life, and technical and vocational skills.

With regard to sports Lesotho will be a leading sporting country. The challenges facing the country include building facilities country wide and strengthening Basotho youth's capabilities in sports.

Lesotho has a potential to be a **strong and prosperous nation** in *macro-economic performance terms*. Its Macro-economic policy has been largely conducive to strong economic growth. In the periods 1970/71-1974/75 and 1975/76-1979/80, the gross domestic product (GDP) grew at an annual average of 8 percent. During 1980-1990 and 1990-1999, growth slowed down, but remained strong, averaging 3.9% and 4.2%

respectively. Since the 1970s, the economy has been transformed from one dominated by agriculture to one dominated recently by manufacturing. While this has improved livelihood options in the urban sector, it has significantly undermined the capacity of the rural and agricultural sector as a source of livelihood, employment and income. The shifting balance towards manufacturing has also worsened poverty in rural communities, particularly those that depend on food production. It will therefore be necessary to take measures to restore the sources of livelihoods for the rural population.

Notwithstanding higher-than-average growth over many years, the challenge at the macro-economic level is to sustain strong investment levels, driven by high domestic saving rates as well as access to international credit markets. This will call for prudent economic and financial sector policies, firstly to promote domestic resource mobilisation and secondly, to retain access to international financial markets. Negotiation of trade agreements and bilateral and multilateral cooperation agreements, membership of regional bodies such as CMA, SACU, SADC and retention of support of development partners are a few key actions that Lesotho has undertaken to integrate into the global economic order.

**Employment creation and prosperity** are of national importance within the macroeconomic performance in Lesotho. Challenges facing Lesotho include the need to address the depth and severity of poverty which increased during the periods 1986/87 and 1994/95.

Lesotho aspires to have a well-managed environment by 2020. The country has signed and ratified several Multilateral Environmental Agreements (MEAS). The spectacular scenery of the Lesotho highlands, the country's unique ecosystem, biodiversity and heritage offer a great potential for the country's tourism opportunities.

Lesotho is however faced with the challenges of implementing the ratified conventions and treaties for sustainable development, strengthening institutions responsible for natural resources management, development and effective implementation of land management systems. A further challenge is to strengthen environmental management, advocacy and awareness among Basotho.

In pursuance of its goal to make Lesotho **a country with a well-established technology**, the Government has, among other things, adopted a privatisation policy to liberalise the telecommunications sector. In this regard, Lesotho is faced with the challenges to move towards a technology competent country through, among others, increased budget allocation towards science and technology development, forging partnerships with other countries, strengthening science and technology education, as well as promoting science and technology research, innovation and development.

Chapter 3 identifies and prioritises strengths, weaknesses, opportunities and threats that should inform strategic choices in Lesotho. The major strengths of the country include the Government's commitment to development, widely accepted and respected constitution, cultural homogeneity, the electoral system and high adult literacy. Major weaknesses on the other hand include food insecurity, high rate of

unemployment, poor strategic and operational planning, inadequate research in science and technology, and an underdeveloped SMME sector. In the external environment the major opportunities are foreign direct investment and good relations with the Republic of South Africa, while the major threats include brain drain, donor conditionalities, decline in mine labour remittances and the increasing competition from international markets.

There is a wide gap between the present situation and the desired vision. For Lesotho to realise its vision there are three major challenges namely: improvement of the development management capacity; sustenance of the investment currently characterising Lesotho's economy; and sustenance of political commitment and support to the Vision up to the year 2020.

The scenario analysis identifies development management capacity, future trends in investment and political stability as critical uncertainties that challenge the Vision. The four scenarios analysed in Chapter 4 are: *Melupe* (Political stability, effective public sector management and favourable trends in investment); *Khomo ha li kae batho re bangata* (Political stability, effective public sector management and declining investment); *Metse e metle liotloana* (Political stability, favourable trends in investment and weak public sector management); and the *Lesupi* (Political instability, weak public sector management, declining investment). The Chapter concludes with an observation that the *Melupe* is the best-case scenario which portrays the desired future for the Kingdom.

Informed by the scenario analysis, Chapter 5 focuses on a two level national development strategy for achievement of Lesotho's Vision 2020. The first level is the **overall or grand strategy,** while **thematic strategies** is the second level which addresses the components of the Vision as described in Chapter 2 and developed as a framework for the development environment scan in Chapter 3. Chapter 5 further clarifies the national development strategy through: a) matrices 5.1 to 5.3 on the strategic actions, monitoring indicators, means of verification and main actors; and b) matrices 5.4 to 5.10 on key strategic actions, monitoring indicators and the main actors for addressing thematic issues.

Implementation, monitoring and evaluation of the Lesotho Vision 2020 are the subject of Chapter 6. The Chapter first identifies strategic actions for the Vision implementation process and depicts challenges that must be recognised therein. The challenges are: a) integrating plans for different time horizons; namely: annual budget statements, three year rolling plans, PRSP, PSIRP and Vision 2020; b) sustaining the strategic thinking process to ensure identification of emerging strategic issues; and c) monitoring and evaluation of the Lesotho Vision 2020.

Chapter 6 also identifies Vision-supporting institutions and their roles. Such institutions include: the Parliamentary Committee on Planning, the National Planning Board and the privately managed and independent Think Tank. Matrices 6.1, 6.2 and 6.3 in Chapter 6 present monitoring indicators of the grand strategy, the Vision components and targets, and key activities for kick-starting the process of implementation, monitoring and evaluation of the Lesotho Vision 2020.



## CHAPTER 1 PURPOSE AND RATIONALE

#### 1.1 Introduction

Since independence, Lesotho has used medium-term planning as the key instrument for coordinating development activities in the country. The six development plans that have been implemented so far had planning horizons that did not go beyond five years. In 2000 the country took a policy decision to formulate Lesotho Vision 2020 to provide a long-term perspective within which short to medium-term plans could be formulated. The specific objectives of the Lesotho Vision 2020 are to:

- establish a long-term vision for Lesotho by looking beyond the short-term plans and adjustment programmes;
- explore the options for economic, political and human development to the year 2020;
- identify alternative development strategies suitable for the Lesotho situation;
- promote a process of open dialogue and consultation with socio-economic groups country wide;
- create an environment in which Basotho will actively participate in achieving the Vision;
- develop a focus along the horizon in the direction of which development plans can be rolled out

#### 1.2 The Process

The process of developing the Lesotho Vision 2020 went through three stages, namely, the inception stage for political commitment and support, the consultation stage and the formulation stage.

#### 1.2.1 Political Commitment

In his call from the Throne on 31 December, 2000, His Majesty King Letsie III called on all Basotho to work together towards developing a Vision to guide the country's economic development in the years to come. "This mammoth task" has become necessary because

"development in all aspects of our life has stalled." Following this call, The Right Honourable the Prime Minister, met with all the leaders of political parties and secured their support for, and commitment to, the vision process.

#### 1.2.2 Consultations

At the directive of the Government, senior government officials developed a framework to initiate the process and lay the foundation for the formulation of a clear vision for Lesotho into the year 2020. Consequently, a three-day National Dialogue was held from 17 to 19 January, 2001, in which all sectors of the Basotho were represented by over 500 stakeholders.

Following the Dialogue, a National Steering Committee (NSC), made up of 40 representatives from a wide spectrum of stakeholder groups, was established and launched in early December, 2001. The NSC's mandate was to develop a National Vision Document informed by wide consultations at all levels of society.

In these consultations, the NSC collaborated with the Technical Working Group for the Poverty Reduction Strategy Paper (PRSP) and held consultations to gain the perspectives of stakeholders at village, ward and district levels. Further consultations were held with political party representatives, chiefs, interim local councils, youth groups, churches, the business community, professionals, marginalised groups, people with disabilities, ethnic groups, civil servants, security establishments, tertiary institutions, primary and secondary schools and children.

#### 1.2.3 The Preparation of the Document

The preparation of the document started in March, 2003. This document is based on, among others, the reports on community and focus group consultations and the National Dialogue. Further, the Vision Drafting Team undertook literature reviews and conducted workshops and interviews with national experts to complement the results obtained from communities and focus group consultations.

#### 1.3 Methodology

The methodology used to formulate this document is a combination of participatory futures studies, strategic planning, and development planning. Participatory futures studies provided the tools and techniques for constructing and analysing alternative scenarios of the futures that Lesotho could face. The tools and techniques of strategic planning were applied to undertake a situational analysis of the development environment and to analyse the strengths, weaknesses, opportunities and threats that should inform strategic choices. Finally, following the theory and practice of development planning, a development strategy was formulated to include the grand and thematic strategies.

The Lesotho Vision 2020 document addresses four basic questions regarding an appropriate long-term development strategy:

How do the Basotho envision their future?

- What are the challenges that face the development process?
- What could happen to the development process in the future?
- What should be done to realize the Lesotho Vision?

The first question relates to the identification of the vision and aspirations of the Basotho. The second question addresses the challenges that face the development process. The third question involves a study of the possible future environment within which Lesotho's development could take place to the year 2020. The fourth question requires that appropriate strategies be formulated to guide public and private sector organizations as well as civil society.

#### 1.4 The Structure of the Document

The rest of the document is structured as follows: Chapter 2 presents the Lesotho Vision 2020; Chapter 3 deals with the development environment; Chapter 4 explores the possible future scenarios; Chapter 5 proposes the strategies to achieve the Lesotho Vision 2020; Chapter 6 addresses implementation, monitoring and evaluation of the vision.



### CHAPTER 2 LESOTHO VISION 2020

#### 2.1 Introduction

This Chapter presents the Lesotho Vision 2020 statement and describes its components. The Vision statement was adopted from the First National Dialogue. The description of Lesotho Vision components is based on the aspirations of the Basotho people. The components are: a stable democracy, a united nation, a nation at peace with itself and its neighbours, a healthy and well developed human resource base, a strong economy and prosperous nation, a well managed environment as well as a well advanced technology.

#### 2.2 The Lesotho Vision 2020 statement

At the first National Dialogue more than 500 representatives of key stakeholders formulated a vision statement as follows:

By the year 2020 Lesotho shall be a stable democracy, a united and prosperous nation at peace with itself and its neighbours. It shall have a healthy and well-developed human resource base. Its economy will be strong, its environment well managed and its technology well established.

#### 2.3 Description of the Vision

#### 2.3.1 A Stable Democracy

By the year 2020, Lesotho shall be a stable democracy where the principles of good governance will be anchored on the respect for human rights, the rule of law, political openness, political participation and tolerance. This form of governance will be based on five pillars of democracy which include: supremacy of the will of the people, transparency, a devoted and efficient public service, justice for all and efficient chieftainship.

In addition, Lesotho will have a well-established system of local governance with full ownership and participation of the majority in decision making and local development. The country will have a clear direction in policy development, strategic thinking and foresight and programme planning. Lesotho will be known for good governance driven by effective

and visionary leadership. Lesotho will be a corruption free country. It's research capacity will be enhanced and information management systems well coordinated.

There will be freedom of expression in all forms of communication without undue interference, and Basotho will have the right to impart and receive information freely. Lesotho shall have a vibrant, free, independent, pluralistic and diverse media regulated by the principles set out in a media policy.

There will be no gender disparities. Men, women, and people with disabilities will be equal before the law; and will be accorded equal opportunities in all aspects of life.

#### 2.3.2 A United Nation

By 2020, Basotho shall be a united nation with cherished norms and values that will enhance a sense of belonging, identity and pride in every Mosotho.

In addition, the Basotho will share common goals based on common cultural heritage and will continue to observe with respect their national symbols such as the national flag. The Monarchy will continue to play the most important role of unifying the nation.

Lesotho will be characterized by sustained political stability, political tolerance and calm political contestations. This state of unity will be a good confidence measure for local as well as foreign investors.

#### 2.3.3 A Nation at Peace with Itself and Its Neighbours

By 2020, Basotho shall be a peaceful nation living in harmony with their neighbours. The nation will have long and lasting peace characterized by truthfulness, love, tolerance, justice, honesty, unity and plenty. Basotho will be a coherent society with a fair distribution of income and wealth.

At the international level there will be peaceful and strong strategic relations with the Republic of South Africa (RSA) and a full proactive involvement with other countries, regional and global institutions. The nation will successfully deal with its challenges including crime, unemployment and poverty. Basotho will have an advanced sense of political maturity and effective conflict and dispute resolution mechanisms to safeguard their peace.

#### 2.3.4 A Healthy and Well Developed Human Resource Base

Basotho shall be a healthy nation with a well-developed human resource base. The country will have a good quality health system with facilities and infrastructure accessible and affordable to all Basotho, irrespective of income, disabilities, geographical location and wealth. Health personnel will provide quality health service and patient care. All Basotho will be conscious of healthy lifestyles and will engage in sporting and recreational activities. Sporting facilities will be available countrywide and Lesotho will also be known as a leading sporting country.

The people will also be highly careful about hygiene and proper nutrition. For all these to be realised, there will be adequate incentives to retain professional staff in the country.

The citizenry will enjoy the benefits of the social protection fund without any discrimination. Traditional and modern medical practices will be fully integrated. All Basotho will have access to safe drinking water and basic sanitation.

There shall be no new HIV and AIDS infections. There will be care and support for the orphans, the HIV infected and affected patients.

Basotho will have access to quality education fully responsive to the country's needs, accessible at all levels and limited only by intellectual ability not by income or wealth. Lesotho will have the system of education that is closely linked and well researched to enhance the student's talents and capabilities. Education will be free and compulsory up to senior secondary level.

Lifelong learning, vocational, technical and entrepreneurial education will be the main focus in the education system. The education system will produce a competent, skilled and productive labour force. In this regard, Lesotho will serve as a service country exporting human capital to other countries whilst retaining a reasonable proportion in the country.

Lesotho will have a dynamic National Manpower Plan geared towards addressing the country's economic, social and political needs. Service delivery as a whole will improve to world standards both in the private and public sectors of the economy. The country will be led by a highly service conscious management and leadership.

#### 2.3.5 A Strong Economy and Prosperous Nation

Lesotho shall be known for the good economic policy that will render the country prosperous for many centuries to come. Its vibrant economy with a strong industrial base will fully exploit the potential for locally available natural resources. The fruit from utilising the natural assets such as water, sandstone and other minerals will be enjoyed by all the citizens. The products of the country will compete well in international markets. Productivity too will be enhanced to world standards.

Opportunities for full employment will be abundant, and Lesotho will be a magnet retaining its people to jobs in the country. The country will, in addition, have well developed Small Micro and Medium Enterprises (SMMEs). A well-developed entrepreneurial culture will be inculcated among the Basotho.

The tourism potential will be fully exploited and the foundation for industrialization well established. Also, characterising the strong economy of Lesotho will be efficiency in tax administration and accountability in the management of government property and national assets.

Lesotho will network and collaborate with other countries in trade, investment and economic advancement in general. The financial intermediation sector will be diverse and highly responsive to customer demands and needs. Access to credit and good loan

management and repayment mechanisms will be the corner stone for development and promotion of the SMME sector and informal sectors.

A Proper economic infrastructure, including roads, telecommunications and electricity networks, well managed and well maintained toll gates and other utilities and facilities will be established. Lesotho will experience a drastic decline in the gap between the rich and the poor. Agricultural production will be at its peak driven by availability of resources (material and human) and proper utilization of agricultural resource centres. The country will then experience a full state of food security and ultimately, commercialized agriculture.

This state of prosperity will be manifested in reduced poverty levels where every Mosotho will afford a basic nutritious meal, adequate shelter and attain a relevant and productive education which will enhance the standard of living.

#### 2.3.6 A Well Managed Environment

Lesotho shall be renowned for its environmental management. The country's diversity of life systems will be supported and protected by a nation which is environmentally conscious and whose people are in balanced existence with the natural environment. Basotho will derive continuing benefits from the conservation and sustainable use of their biological diversity. The several global conventions and treaties that Lesotho has signed and ratified shall be translated into concrete actions which will sustain care and management of the environment at large.

Basotho will be empowered in designing and managing biodiversity conservation projects relevant to their own communities. Environmental education will be integrated at all levels of learning. There will be institutional and legal frameworks to promote and protect a healthy and sustainable environment. Every development in the country will be subjected to an intensive Environmental Impact Assessment (EIA) to gauge its environmental friendliness.

The country's urban areas will be exemplary in proper structural and infrastructural networks intended to ease life. In terms of shelter, people will observe regulations and requisite building standards, and will make optimal use of available space for housing in the urban areas. The spectacular scenery of the Lesotho highlands, the unique ecosystem and a wide biodiversity and heritage will render the country a good tourist destination.

#### 2.3.7 A Well Advanced Technology

Lesotho will have a solid foundation for technological research and development. The school curricula shall be strong in science and technology which will lead to increased enrolment in sciences and technical programmes. The nation will expand its usage of applied technology in all aspects of life. Ninety percent of Basotho households will have access to electricity and thereby to communication and development technology.

The country will increase its financial and human resource allocation towards innovative research. This will enable the Kingdom to reduce the technological divide between herself and the developed countries. Local production and diffusion of appropriate technology will be widespread to meet the high demand. Furthermore, application of appropriate

production technology as well as imported technology will increase in all spheres of Basotho's life and thereby contribute to the reduction of poverty.

#### 2.4 Conclusion

The chapter has presented the Lesotho Vision 2020 statement and described its components. To realize this vision, all Basotho people need to have a unity of purpose, well thought-out strategies and a determination to succeed.



# CHAPTER 3 THE DEVELOPMENT ENVIRONMENT

#### 3.1 Introduction

This Chapter reviews trends in the development performance of Lesotho. The review identifies key trends, events and challenges in different sectors of the economy, looking at both the internal and external environment. The section further seeks to identify and prioritise the strengths, weaknesses, opportunities and threats that should inform strategic choices in Lesotho.

Lesotho's external environment comprises its membership in international organizations, multilateral and regional agreements to which it is party, international market opportunities and Official Development Assistance (ODA). Lesotho is also a member of the United Nations (UN) and the African Union (AU). It subscribes to the principles of New Partnership for Africa's Development (NEPAD) which is a pledge by African leaders, based on a common vision and a shared conviction to eradicate poverty and sustain growth and development of their countries.

The multilateral arrangements include the Common Monetary Area (CMA), Southern African Customs Union (SACU) and the Southern African Development Community (SADC). The international market opportunities include African Growth and Opportunity Act (AGOA) and the ACP-EU Cotonou Agreement.

#### 3.2 Assessment of the Development Performance

This section reviews the key aspects of the internal environment as they affect national development prospects. The analysis is anchored on the components of the Lesotho Vision 2020 Statement adopted at the First National Dialogue<sup>2</sup> and explained in detail in Chapter 2 of this report.

- Nation at peace with itself and its neighbours = (peace)

 $<sup>^{2}</sup>$   $^{2}$  The components of the Vision Statement have been clustered as follows;

<sup>-</sup> Stable democracy = (governance, media, gender)

<sup>-</sup> United nation = (culture)

<sup>-</sup> Healthy and well developed human resource base = (health, HIV and AIDS, education, sports)

Strong economy and prosperous nation = (economy, prosperity)

Well managed environment = (environment)

<sup>-</sup> Well established technology = (information, communication, science, technology)

#### 3.2.1 A Stable Democracy

Good governance, media freedom, and gender sensitivity characterise Lesotho's stable democracy.

#### a) Governance

Lesotho is a democratic country with the Monarch as head of state, the Prime Minister as head of government. Although the Kingdom's democracy has been punctuated with political instability and disturbances, efforts have been made to bring the situation under control. For example the end of the military rule in 1993 saw a democratic era which heralded the establishment of institutions of governance such as the Independent Electoral Commission (IEC) and the Office of the Ombudsman, in addition to the practice of separation of powers between the three arms of government, namely, the Executive, the Judiciary and the Legislature.

While there is a considerable respect for the rule of law, there are disconcerting delays in the dispensation of justice. Governance in Lesotho has been strengthened by the recently adopted electoral system that has for the first time in the history of the country enabled ten political parties to be represented in parliament.

The challenges that the Kingdom still has to contend with in relation to governance are: empowering the legislature to carry out the oversight function, improving the Judiciary to deal with the delays in the justice system and, strengthening the public service for delivery of services and accountability. The other challenge is to strengthen and professionalise the security establishments.

Development management capacity together with local governance and popular participation are aspects of governance within a stable democracy.

#### i) Development Management Capacity

In addition to the regular five-year national development plans which have recently been changed to three year rolling plans, the country has also developed comprehensive long-term plans, namely, *Action Programme for Development of Lesotho 2001-2010, The Strategic Options study;* and *The Pathway out of Poverty.* 

Lesotho has embarked on the Public Sector Improvement and Reform Programme (PSIRP). This programme represents the Government's framework for public sector improvement and reform in: Good Governance, Financial Management and Budgeting, Human Resources Management, Administration of Justice, Restructuring and Rationalisation of Ministries, Departments and Agencies, Decentralisation and Local Government, and Delivery of Services. The first phase of the programme focuses on three areas, namely, improving financial management and accountability, improving public service management, and decentralisation for service delivery.

Furthermore, the country has developed a Poverty Reduction Strategy Paper (PRSP) and developed a Medium Term Expenditure Framework (MTEF) for development and financial management.

The Constitution of Lesotho provides for the establishment of the National Planning Board to: prepare plans for economic development, coordinate and supervise the preparation of such plans by the Government and other public authorities, and advise the Government and other public authorities in relation to economic development.

The key challenges in development management capacity include: improving research capacity, coordinating information management systems, brain drain in different sectors, implementation of PSIRP, strengthening of public, private and civil society partnership in development, corruption, nepotism and defining roles and responsibilities between the National Planning Board and the Ministry of Finance and Development Planning. Another key challenge is to improve management of the donor funded projects and strengthen the absorptive capacity, reporting and coordination. Development of a National Aid Policy should be another priority area.

#### ii) Local Governance and Popular Participation

The country is working towards decentralization by implementing the Local Government Act of 1997. It is envisaged that local government elections will be held in 2004 to increase participation by the ordinary citizens and the rural poor in the governance process. This will enable individual citizens and civil society in general to contribute to the development process.

The challenge in this regard is to empower the imminent local government authorities, and to improve chieftainship as a strategy to complement governance at the grassroots level.

#### b) Media

Media in Lesotho is free from direct government interference. This is supported by the pluralist nature of the media in all its forms, which reflects diverse opinions of people from all walks of life. In spite of the proliferation of media houses, media diffusion with all its components, namely, print and electronic, is predominantly in urban areas. The national radio station (Radio Lesotho) is the only one that has nation-wide coverage. However, plans are underway for increasing television coverage.

The challenges here include the consolidation of the currently scattered pieces of legislation regulating the media, inculcating media professionalism and independence, and increasing media coverage to the rural areas of the country.

#### c) Gender

Lesotho's female population enjoys higher education attainment and literacy than male population. This enables upward mobility for them in various sectors of the economy. Moreover, there is a clear policy and legislation on gender.

Government established a Law Reform Commission in 1997 to review all laws that are discriminatory, in conflict with the Constitution, or outmoded. As a result, laws that discriminate against women are being dealt with accordingly. The civil society also plays a major part in gender issues through advocacy and influence in law enactment.

In pursuance of gender equity and equality among the Basotho society, Lesotho has signed and ratified the Convention on Elimination of all forms of Discrimination against Women (CEDAW), albeit with reservations on certain sections that have constitutional implications regarding customary laws, the church and chieftainship. It is also party to the SADC Addendum on the Prevention and Eradication of Violence against Women and Children signed in 1997.

The key challenge is to uproot discrimination and appoint more women into areas of responsibility and decision making in both the public and private sectors. Another challenge is to uplift women without neglecting boys and men.

#### 3.2.2 A United Nation

Culture and the role of the Monarchy are pivotal to national unity in Lesotho. The role of the Monarchy is to play a unifying role which is important for maintenance of peace and unity in the country. The system of governance that puts the King above party politics has enhanced his status as a symbol of unity for the nation.

Culture is central to national unity in Lesotho. The widespread use of national symbols such as the national flag signifies a united nation sharing common goals based on a common cultural heritage. The Ministry of Education and Training encourages cultural appreciation in curricula development, and consciously promotes cultural practices through a mandatory cultural day in schools. Most initiatives in the revival and promotion of cultural practices are taken by private institutions with the support of the Ministry of Tourism, Environment and Culture, through annual arts and cultural festivals. In a nutshell, Basotho's common history, common language and traditional practices all contribute to towards homogeneity which in turn also promotes national unity.

The greatest challenge is to inculcate the spirit of patriotism in all Basotho. Other challenges are protection of Lesotho's culture and heritage resources, improvement of facilities for artists and their product market, research and documentation of Basotho cultural practices.

#### 3.2.3 A Nation at Peace with itself and its neighbours

Basotho are a peace loving nation. This peacefulness is embedded in the greeting "Khotso"- meaning 'Peace'. Their peacefulness is reinforced by the fact that they are a

nation with Sesotho as the main language and a shared history, culture and traditions. The new electoral system which is a combination of first-past-the-post and proportional representation has brought about more political party representation in parliament and, thereby more political calm and peace.

However, a number of challenges have to be addressed. Firstly, there is a need to combat the current rate of internal and cross-boarder stock theft, crime and armed robbery which deprive communities of the necessary state of peace. The second challenge concerns improvement of political tolerance, conflict and dispute resolution mechanisms.

#### 3.2.4 A Healthy and Well Developed Human Resource Base

Health, HIV and AIDS and education are critical issues to be addressed in a country with a healthy and well-developed human resource base.

#### a) Health

Lesotho adopted the Primary Health Care Strategy (PHC) in 1979. The strategy encompassed national immunisation and nutritional programmes. The PHC along with other strategies led to the significant improvement in the health survival indicators related to life expectancy at birth, infant, child and maternal mortalities through to early 1990s. The country has also made remarkable progress in providing access to social infrastructure, including access to safe drinking water and sanitation. Between 1990 and 2001 the proportion of the population without access to safe drinking decreased from 66% to 55%. The proportion of the population without access to safe drinking decreased from 36% to 23% in the same period.

To reverse the recent erosion of the positive trends in the survival indicators, particularly as a result of, among others, HIV and AIDS, poverty and other socio-economic difficulties, the country faces a number of challenges. These include equitable distribution of health facilities to all sectors, and strengthening the health institutions for efficient and effective service delivery. Furthermore, the sector needs to develop a comprehensive health policy framework.

#### b) HIV and AIDS

At the end of 2001, approximately 30% of the population was infected with HIV and AIDS. The Government of Lesotho recognizes that HIV and AIDS is not only a health problem but a multi-sectoral development issue that has social, economic and cultural implications. In 2000, the Government in collaboration with multi-sectoral stakeholders developed a National Policy on HIV and AIDS Prevention, Control and Management to create a conducive environment for prevention of further spread of HIV and AIDS and to mitigate the adverse impact on the infected and affected individuals, families and communities.

The government of Lesotho has also committed 2% of its ministerial budget allocations towards HIV and AIDS prevention and impact mitigation programmes. Donor assistance also plays an important role in this regard through provision of financial, material and technical resources as well as technical assistance.

Combating further spread of HIV continues to be one of the biggest challenges that face the country. Effective implementation of the national policy and strategy which include prevention, mitigation, care, support and treatment of the infected and affected would go a long way in this regard.

#### c) Education

Lesotho's adult literacy rate (82%, 2002) is higher than it is in most African countries. The country is committed to the provision of an equitable basic education to all Basotho as a key development goal. Introduction of Free Primary Education (FPE) Policy in 2000 together with other sectoral and sub-sectoral policies provide hope for every Mosotho to have access to basic education and in turn a better future. Lesotho's education system is a joint responsibility of the Government, civil society and the churches as strong partners. Lesotho is also a signatory to a number of international and regional conventions and declarations to which it fully complies.

The education system is yet to fully respond to the needs of the country. Key challenges include: further strengthening of access to education, developing curriculum that fully responds to the national development priorities, thus, promoting entrepreneurial, life, technical and vocational skills. Lesotho has to develop education system that is universal, free and compulsory to all. The present system of examinations needs further improvement to avoid potential anomalies. Improvement of management capacity of the education institutions at all levels and the reversal of the adverse effects of the HIV and AIDS pandemic on the education system should be given priority.

#### d) Sports

In 2002, the Lesotho Government passed the Lesotho Sports Act which allowed for the formation of the Lesotho Sports and Recreation Commission (LSRC) that was elected in April, 2003. There are presently twenty six (26) different sporting associations affiliated to the LSRC. Lesotho also boasts the presence of the Lesotho National Olympic Committee (LNOC). Lesotho participates in regional and international sporting events including: The Zone VI Games, the All Africa Games, the Commonwealth Games and the Olympic Games.

The sporting challenges facing the country include: building of facilities countrywide, constructing facilities of international standard, financing of the sporting activities and promoting sport tourism. Another challenge is to widen the funding base for sporting activities to include the private sector.

#### 3.2.5 A Strong Economy and Prosperous Nation

Macro economic performance, the structure of the economy, employment, SMMEs and the informal sector and prosperity are central to understanding this component of Lesotho's Vision.

#### a) Macro Economic Performance

Economic growth, fiscal developments, and public debt depict macro-economic performance in Lesotho's strong economy.

#### i) Economic growth

The macro economic policy environment in Lesotho has been conducive to economic growth. In the periods 1970/71 - 1974/75 and 1975/76 - 1979/80, Gross Domestic Product (GDP) grew at an average of about 8% due to good agricultural sector performance, diamond mining during the late 1970's, large volumes of external assistance, increase in migrant labour remittances and the Southern African Customs Union (SACU) revenues.

During 1980-1990 and 1990-1999 periods, GDP grew at an average of 3.9% and 4.2% per annum respectively as a result of the construction phase of the Lesotho Highlands Water Project (LHWP) and the rapid expansion of the manufacturing sector. In 2002, the economy registered positive growth of 3.8% from the 3.2% experienced in 2001 as a result of the rapid expansion of clothing exports through African Growth and Opportunities Act (AGOA).

However, the winding down of some LHWP activities, reduction in miner's remittances and the SACU revenues, 1998 political upheaval and decreased volumes of external assistance have resulted in the recent decline in the economic growth rate.

The challenge is to sustain high economic growth rates through increased national productivity, exploring the alternative sources of revenue and generating job opportunities to absorb the labour force.

#### ii) Fiscal Developments

Lesotho adopted and successfully implemented three-year Structural Adjustment and Enhanced Structural Adjustment Programmes to correct fiscal imbalances experienced in the mid 1980s. These imbalances included widening government budget deficit, due to increased recurrent expenditure and decline in revenue.

Fiscal deficits were also experienced from the late 1990s to 2002/2003 due to the: political turmoil of September 1998, restructuring process of key government institutions, repayment of the commercial loans for the Lesotho Highlands Water Project (LHWP), recent government spending on famine relief and agricultural support.

Recent developments to improve revenue collection include establishment of the Lesotho Revenue Authority (LRA) which aims at managing the Government's revenue policy. The Value Added Tax (VAT) was also introduced in July 2003 as an effective mechanism for tax management.

Lesotho faces a challenge to develop prudent fiscal strategies for good financial management and maintenance of sustainable fiscal balances.

#### iii) Public Debt

Between 1998 and 2001 Lesotho's external debt increased sharply (from 68% in 1998 to 108.2% in 2001) due to a sudden depreciation of loti against the major borrowing currencies. However, in 2002 the external debt declined by 17.9% mainly because of the appreciation of the local currency and also the retirement of the loans which were costly.

Government accounts for an average of 90% of the outstanding public debt and the remaining 10% is shared among other public entities. Approximately 80% of total external debt has been provided by multilateral lending agencies, principally the World Bank and African Development Bank. It is a challenge for Lesotho to sustain internationally accepted prudent levels in terms of debt service ratio.

#### b) Structure of the Economy

#### i) The Primary Sector

Approximately 75% of the Lesotho population live in the rural areas and derive their livelihood from agriculture. In 1966 the share of agriculture in the GDP was about 50%. In recent years it has declined to below 20%.

The negative performance in the sub sector is attributable to, among others, a drop in crop production due to frequent droughts, heavy frosts and late rains experienced in recent years. The fact that 9% of total land area in Lesotho is arable puts pressure on the fertility of a limited land base. The Agricultural sector has also depended heavily on external assistance. With the general decline of donor support in the country, the sector became the hardest hit. The average agriculture share of total development assistance declined significantly.

The sector is however expected to show some recovery with implementation of the Agricultural Sector Investment Programmes. The challenge is for Lesotho to increase agricultural productivity to sustain food security in the country.

#### ii) The Secondary Sector

Lesotho's performance in attracting Foreign Direct Investment (FDI) is well above average for low and middle-income countries. Through FDI flows, Lesotho has been able to take advantage of trade privileges in US markets under the Multi Fibre Agreement (MFA) and AGOA starting April 2001.

Table 3.1 below shows that the volume of FDI to Lesotho increased significantly from an average of USD 213 million in 1990 to 1995 to a peak of US\$286 million in 1996. Since then FDI has declined steadily to US\$118 million in 2001, representing a 34.3 percent decrease from the 1990 to 95 level.

Table 3.1 Lesotho: Selected indicators of FDI

Indicator	1990-1995 Average	1996	1997	1998	1999	2000	2001
FDI Inflows (Millions of dollars)	213	286	269	262	163	119	118
FDI inflow as % of gross fixed capital formation	44.4	52.0	47.8	60.2	37.7	31.1	N/A*

Source: UNCTAD: World Investment Report 2002. Annex Table B.1

The construction industry is also expected to perform moderately well through private investment and expansion of factory shells. The government is currently establishing more industrial sites at Tikoe and Mohales'hoek.

This sector is, however, faced with a number of challenges that the country needs to contend with. The preferential market opportunities to which Lesotho exports its products are temporary measures due to expire soon. For example, the AGOA will expire in 2015 with the third country fabric sourcing benefit due to expire by September, 2007. Lesotho has to develop its capacity to fully take advantage of the presence of foreign owned firms in the manufacturing industry to forge backward and forward linkages with the local businesses. The industrial base needs to be widened to diversify product and market outlets. This will reduce the potential vulnerability to adverse developments in the two main markets; namely, the Republic of South Africa and the United States of America. Furthermore, Lesotho still has to do a lot more to create an environment to attract nationals and foreigners to invest in the country. Licensing procedures, tax regime and land acquisition and transfer need further review to facilitate more investment.

#### iii) The Tertiary Sector

Much of the growth reported in the wholesale and retail sub-sector is attributable to electricity, telecommunications and motor vehicle sales. Growth in this sub sector continued to give a clear indication that the economy is indeed benefiting from the reform of the utility industries initiated in 1999. The tertiary sector is projected to grow by an annual average of 3.6 %. This will be encouraged by continued expansion and investment in value added by the Post and Telecommunications sub-sector. Expenditure on health and education are expected to grow in line with the Government's commitment to achieve the Millennium Development Goals (MDGs) in these areas.

The challenge in this sector is to implement recent institutional reforms that will lead to improvements in the performance of the Tourism sector as demonstrated by potential in this sector.

#### c) Employment

Employment creation remains a national priority. Prior to 2001 the main source of employment in the domestic economy was the Government. Government employment

(comprising civil servants, teachers and the armed forces) in 2002 increased by 0.7% to 36,038. In 2001, the introduction of the AGOA initiative changed the trend with the manufacturing sector gaining prominence. Employment generated in the manufacturing sector is estimated at around 44,537 (2002).

Unemployment in Lesotho is estimated at over 40% of the labour force. This has been caused by among others, the diminishing employment prospects in the agricultural and government sectors, the small industrial base and the retrenchment of Basotho mineworkers from the Republic of South Africa. Lesotho has relied heavily on attracting foreign investment mainly in textile manufacturing and to some limited extent in the construction industry to curb the problem of unemployment.

High economic growth rates of the 1990s did not translate into substantial employment opportunities. The challenge facing Lesotho is to generate growth with sustainable employment opportunities for every Mosotho, both skilled and unskilled. Promotion of more labour intensive sectors and development of comprehensive employment policy framework are areas of priority in the fight against unemployment.

#### d) SMMEs and the Informal Sector

The unemployment challenges facing the country today have compelled the Government to attach great importance to the development of SMMEs and the informal sectors. Recent estimates reveal that the sector employs over 130,000 people making it the largest employer.

However, there has to be adequate and sustainable provision of business services, proper infrastructure, availability of capital and access to markets and credit. The business environment with regard to licensing is too complex and inefficient and the tax regime still favours manufacturing at the expense of the service sectors. The sector has been without clear National Industrial and SMME Policies for a long time. Land acquisition and transfer remain a complex and discouraging investment. As a result women remain legally restricted from becoming independent economic agents.

The challenge is therefore to develop this sector for it to make a meaningful contribution to the economy.

#### e) Prosperity

The UNDP Human Development Report 2002 classifies Lesotho among the medium human development countries and ranks it 132nd out of the 173 countries on the Human Development Index. In 2000 Lesotho ranked 127<sup>th</sup> out of 174 countries.

The Bureau of Statistics-University of Natal study based on the 1986/87 and the 1994/95 Household Budget Survey (Table 3.2) revealed that the percentage of people living below the national poverty line remained unchanged at around 58% between 1987 and 1995. While the incidence of poverty improved marginally between 1986/7 and 1994/5, the depth and severity of poverty increased during the period as Table 3.2 below shows.

Table 3.2 Incidences, Severity and Depth of Poverty (1986/7 and 1994/5)

	Poverty	Line (PL)	Ultra Poverty Line (1/2 PL)			
	1986/7	1994/5	1986/7	1994/5		
	(%)	(%)	(%)	(%)		
Incidence	58.8	58.3	34.7	38.6		
Depth	32.8	35.4	17.7	21.4		
Severity	22.8	25.9	11.8	14.9		

Source: Bureau of Statistics – University of Natal estimates based on the 1986/87 and the 1994/95 Household Budget Surveys

The key characteristics of the inequality are the rural/urban divide. The rural areas suffer more income poverty than the urban areas. The Human Development Report 2002 reports a Gini index<sup>3</sup> of 56.0 for the survey year 1986-87. The share in income of the poorest 10 percent was 0.9 percent while the richest 10 percent had a share of 43.4 percent.

The average household size in Lesotho is 5.0 and 49% of Lesotho's population is young children below the age of 14 with the elderly population (65 and over) estimated at 5.8%. This youthful age structure together with prevalence of HIV and AIDS results in high dependency ratio.

The key challenge for Lesotho is to bridge the gap between the poor and the rich through improving the distribution of income and wealth among Basotho. Improved food security, combating further spread of HIV and AIDS, improved access to social infrastructure should be the priority sectors for building a prosperous Lesotho. Successful implementation of the Poverty Reduction Strategy Paper provides a hope for adoption of effective strategies for poverty reduction.

#### 3.2.6 A Well Managed Environment

Lesotho has signed and ratified several Multilateral Environmental Agreements (MEAs). The National Environmental Action Plan (NEAP) was formulated in 1988 and updated in 1999 to integrate environmental issues into economic development. The Environment Act was enacted in 2001 to provide an overall legal and regulatory framework for setting environmental standards, Environmental Impact Assessment (EIA), procedures and compliance for the agreements. The implementation of this Act will begin as soon as all necessary structures such as Lesotho Environment Authority have been established.

The spectacular scenery of the Lesotho highlands, the country's unique ecosystem, biodiversity and heritage offer the country tourism opportunities and great potential. The Ministry of Tourism, Environment and Culture collaborates with the newly established

<sup>&</sup>lt;sup>3</sup> The Gini index measures the extent to which the distribution of income (or consumption) among individuals or households within a country deviates from a perfectly equal distribution. A value of 0 represents perfect equality, a value of 100 perfect inequality.

Lesotho Tourism Development Corporation (LTDC) to harness this potential. Renewable energy technologies upon which the majority of Basotho people depend and which are environmentally friendly, supply very little energy in the country. Lesotho is endowed with rich water sources and water is one of the revenue earning resources.

Lesotho is however faced with the challenges of: effecting the ratified conventions and treaties for sustainable development, strengthening institutions responsible for natural resources management, development and effective implementation of land management systems, controlling harvesting of natural resources, and improvement of effluent and waste management.

#### 3.2.7 A Well Established Technology

Lesotho's draft science and technology policy is based on a vision "to sustain a free, prosperous and progressive economy and society that is sustained through intelligent use of science and technology assisted by enlightened citizens, corporations and government." The Government of Lesotho has adopted a privatisation policy to liberalise telecommunications sector. In 2000, The Lesotho Telecommunications Authority (LTA) was established to promote competition within the telecommunication sector, protect consumer rights, advance Universal Service/Access and manage the electromagnetic spectrum.

Since the liberalization of the telecommunication sector the penetration of telebureaus and payphones has increased almost fivefold. The amounts communicated to Science and Technology (S&T) in the government's budget rose from M812,200.00 in 1997/98 to M2,034, 180 in 2001/02.

In terms of Technology, Lesotho is faced with the challenges of increasing its budget allocation towards science and technology development, forging partnerships with other countries, strengthening science and technology education, and promoting science and technology research, innovation and development.

The number of radio stations has increased from one to eight. There are two television stations and newspapers have also increased to ten.

#### 3.3 Strengths, Weaknesses, Opportunities and Threats (SWOT)

For Lesotho to achieve its Vision the country has to capitalize on its strengths. These include a widely accepted and respected constitution; government commitment to development which is manifested in the formulation and implementation of development plans; cultural homogeneity, and political stability that has been brought about by the new electoral system. As a developing country Lesotho also has limitations that offset its strengths. These weaknesses relate to poor strategic and operational management of private, parastatal and public institutions; inadequate science and technology research and development; underdeveloped SMME and informal sectors; high unemployment rates; food insecurity; and increasing HIV and AIDS prevalence.

The external environment presents the country with opportunities on the one hand and threats on the other. Foreign Direct Investment, strong donor support, good bilateral

relations and multilateral arrangements are some of the opportunities that Lesotho can exploit. Major threats that the Basotho have to contend with are, *inter alia*, increasing competition from the international markets, donor conditionalities, relentless brain drain; retrenchment of Basotho from the Republic of South Africa and declining SACU revenues.

A tabular presentation of the analysis of Lesotho's strengths, weaknesses, opportunities and threats is attached as Annex 3. The analysis is based on critical success factors outlined in Annex 1.

For Lesotho to find a strategic fit between its strengths and opportunities on the one hand and its weaknesses and threats on the other the country has to address some of its key challenges.

#### 3.4 Summary of the Challenges

The following challenges have been gleaned from the above assessment of the development environment:

- Strengthening the three arms of government to deepen the roots of democracy;
- Decentralizing services and power to empower communities at the grassroots;
- Professionalizing media to effect its pluralistic nature;
- Strengthening development management capacity in the country;
- Addressing the issue of corruption and nepotism;
- Supporting and maintaining the principles of Gender equality;
- Instilling the spirit of patriotism among all Basotho for socio- economic and political advancement of the country;
- Preserving and maintaining peace and tranquillity for the country's stability;
- Improving access to health services;
- Implementing National AIDS Policy effectively for national development;
- Improving access to education and training to ground the foundations for good human resource base;
- Improving and strengthening sporting capacity and capabilities;
- Maintaining macro economic stability for economic development of the country;
- Improving expenditure management and sustain sustainable fiscal balances;

- Improving the financial sector through encouraging more competition in the banking, capital markets and insurance sectors;
- Improving & sustaining food security for sustainable prosperity;
- Sustaining high levels of investment for economic up-liftment of the country;
- Strengthening the national capacity to fully take advantage of the presence of foreign owned manufacturing enterprises;
- Reducing the potential vulnerability to negative developments in the two main markets; namely, the Republic of South Africa and the US through Diversifying preferential trade and investment market opportunities;
- Improving the investment environment;
- Finding alternative sources of revenue while strengthening capacity within SACU;
- Negotiating extension of AGOA and MFA agreements;
- Generating growth with sustainable job opportunities to curb the problems of unemployment;
- Improving service delivery in all sectors of the economy;
- Tapping the potential of Tourism effectively;
- Developing and promoting SMME sector;
- Bridging the gap between the rich and the poor for equitable distribution of income and wealth;
- Reducing the incidence, severity and depth of poverty in the country;
- Managing and sustaining healthy environment for sustainable development;
- Promoting production and use of science & technology;
- Identifying alternative sources of revenue to maintain sound fiscal balance in the economy;
- Improving co-ordination of donor funded projects.



# CHAPTER 4 SCENARIO ANALYSIS

#### 4.1 Introduction

In assessing the development prospects of Lesotho in the long-term it is important to forecast the environment within which the development process will evolve in future using futures studies and scenario analysis. Short-term forecasting methods are not effective over a period as long as 20 years.

The situation analysis conducted in Chapter 3 reveals that three critical uncertainties for Lesotho's future development prospects are: development management capacity, the future trends in local investment in the country and the prospects of export markets, and political stability.

The uncertainties form the basis for mapping out possible pathways that Lesotho could follow to the year 2020. Annex 2 identifies the focal issue, and its driving forces, factors behind the driving forces, key trends and uncertainties. It then indicates four scenarios mapped out in different combinations. Scenarios are consistent, but different plausible stories of how the future could unfold depending on a number of factors. Many possible future scenarios for Lesotho were considered, but the four that have been chosen were found to be the most plausible.

#### 4.2 Four Scenarios

In forecasting the environment within which the development process could evolve in Lesotho's future, the following scenario combinations have been selected:

- **Scenario 1:** Political stability, effective public sector management and favourable trends in investment (the *Melupe* Scenario).
- **Scenario 2:** Political stability, effective public sector management and declining investment (the *Khomo ha li kae batho re bangata* Scenario).
- **Scenario 3:** Political stability, favourable trends in investment and weak public sector management (the *Metse e metle liotloana* Scenario).

**Scenario 4:** Political instability, weak public sector management, declining investment (the *Lesupi* Scenario)<sup>4</sup>.

## The "Melupe" Scenario

This scenario considers a situation where there is effective public sector management, sustained political stability and commitment and positive trends in investment.

The effectiveness of public sector management is seen through the implementation of the Vision, demonstrated by effective structures on the ground for implementation. An important manifestation of this public sector management is the effective implementation of the Public Sector Improvement and Reform Programme (PSIRP) which results in an enhanced national capacity to manage development programmes, formulate development policies, and facilitate implementation of private sector investment programmes. Furthermore, technical and other forms of assistance from the international community increase as a result of sound management and effective implementation of development programmes.

Sustained political stability is evident through successful general elections and peaceful transfer of power from one party to another, increasing representativeness of Parliament, and professionalism and loyalty of the armed forces. The country is able to bring crime, nepotism and corruption under control, which in turn improves the delivery of services to the citizenry. As a result of political stability and good public sector management, the Kingdom develops and maintains strategic relations with the Republic South Africa (RSA).

In addition, the Vision becomes a cross-cutting issue fully integrated into the programme appraisal and budgeting procedures of government. Therefore, it becomes mandatory for all government sanctioned development projects and programmes to show how they contribute towards the ultimate achievement of the Vision. This arrangement works very well because public sector management has improved significantly due to the effective implementation of performance management and effective implementation of strategic plans by different government ministries. This enhances implementation of sectoral

<sup>&</sup>lt;sup>4</sup> The four names of the scenarios are derived from Sesotho proverbs that are explained below:

 <sup>&</sup>quot;Melupe" denotes a drizzle, a kind of rain that seeps perfectly into the soil and is attributed to good harvest.

<sup>2. &</sup>quot;Khomo ha li kae batho re bangata" means that resources are scarce and as such there is competition over them. This competition is within and outside the country. Externally, the main resources that countries compete for in this regard are foreign direct investment and market access.

<sup>3. &</sup>quot;Metse e metle liotloana" is a Sesotho proverb which points to the fact that things might look good from a distance, but at a close range be far from being good.

<sup>4. &</sup>quot;Lesupi" denotes total destruction. "Ntlo e itoants'ang e fetoha lesupi" means that when there is instability and lack of harmony, development hardly takes place. Instead there is suffering, total destruction and reversal of economic gains.

strategies and policies. Furthermore, the strategic partnership between the public and the private sector has gone a long way in improving the country's performance in this regard.

A stable and positive environment helps to position the country to reap maximum benefit from favourable FDI inflows and growing external markets. Through local skills and institutional development, the country increases its own capabilities in manufacturing, services and agriculture and is able to increase local investment through increased spin-off effects from the large-scale export oriented companies to the indigenous small-scale entrepreneurs. Lesotho is able to diversify its export products and markets and exploit fully its complementaries with the economy of RSA. This results in a long-term GDP growth rate of 6-7 percent.

In this scenario the government, civil society, the private sector and the nation at large, recognise the importance of human development in all its facets for the effective implementation of the Vision. Policies to this effect are formulated and implemented with enthusiasm and diligence. With the development of human resources, the formulation of appropriate policies, strengthening of institutions to ensure effective Vision implementation and sustained respect for democratic institutions, come good governance. Such governance together with good public sector management results in better management of Lesotho's relations with RSA whose cooperation is critical to Lesotho due to her geographical position. Commitment to the Vision takes cognisance of the importance of preservation of the environment for sustained long-term national development.

Visionary political leadership that the country has under this scenario galvanises nation-wide support for the Vision through, *inter alia*, recognition of Basotho cultural heritage, the sense of national uniqueness, which ultimately conjure strong feelings of nationhood and patriotism. This is the *Melupe* Scenario - the best case achievable with commitment and hard work.

# The "Khomo ha li kae batho re bangata" Scenario

This scenario considers a situation where there is sustained political stability and commitment, effective public sector management and a decline in foreign direct investment.

Despite political stability and improved public sector management, the country experiences a decline in foreign direct investment. Access to foreign markets for the Kingdom's exports is also curtailed. These come as a result of the expiry and non-renewal of favourable trade agreements. Docile labour and low wages in some less developed countries enable them to out-compete Lesotho in attracting foreign direct investment. In spite of the Kingdom's efforts in trying to woo investors and negotiate access to markets, there is a decline in investment. The declining investment and limited access to foreign markets virtually isolate the country in a rapidly globalising world and a closely integrated community of nations.

The *Khomo ha li kae* Scenario results in average GDP growth rates of 2% per annum which really does not make much impact on the heightened state of poverty and unemployment in the country.

#### The "Metse e metle liotloana" Scenario

In this scenario there is political stability, favourable trends in investment and weak public sector management. The ineffectiveness of public sector management is manifested in low morale and lethargy among civil servants. This results in poor service delivery.

Public sector programmes that are meant to facilitate the implementation of the Vision such as the Public Sector Improvement and Reform Programme (PSIRP), Poverty Reduction Strategy Paper (PRSP) and Performance Management System are also not implemented with the requisite vigour and enthusiasm.

The country has a great tourism potential which is not being fully exploited. Drought wreaks havor of the country in spite of perennial water sources due to poor damming and harvesting of water resources. HIV and AIDS is rampant and the Basotho people are dying in great numbers but are seemingly failing to change their sexual behavioural patterns.

Despite sustained political stability and support for the Vision, the country is unable to bring crime and corruption under control which in turn impoverishes people even further and compromises public safety and service delivery. Because of political stability, local and foreign direct investment coupled with access to international markets initially increase and the textile manufacturing industry continues to be the main employer. However, over time, the incremental rate of FDI starts declining because of poor public sector management.

The Republic of South Africa and the Kingdom of Lesotho have signed a strategic agreement – the Joint Bilateral Commission of Co-operation (JBCC). The relations are, however, threatened by the heightened cross-border crime and lack of adherence to some regional arrangements by some operatives in the Republic of South Africa.

The *Metse e metle liotloana* scenario depicts the current (2002-2003) state of the country which could go on for the next ten to fifteen year period. The scenario results in average GDP growth rate of 2.5-3% per annum which does not make much impact on the widespread poverty and high unemployment levels.

# The "Lesupi" Scenario

In this scenario, the situation in Lesotho is characterized by political instability, lack of political will to implement Lesotho Vision 2020, weak public sector management, and negative trends in investment.

Slow implementation of the Poverty Reduction Strategy deprives the country of the opportunity to reduce inequality in the distribution of assets and opportunities. In the face of mounting unemployment and deepening poverty there are agitations for 'quick-fix' solutions to national development problems. Some political parties see this situation as an opportunity to increase their support base by offering short-term development approaches that are unsustainable and contrary to the thrust of Lesotho Vision 2020.

These problems result in lack of political commitment to Lesotho Vision 2020, which leads to a decline in political will for its implementation. All vision-supporting institutions become dysfunctional.

Furthermore, the PSIRP is shelved for lack of political will for implementation. As a result higher levels of individualism, avarice, self-enrichment, corruption and ineffectiveness characterize the public service. The deteriorating public sector management adversely affects management of the Kingdom's relations with other countries in general and those with RSA in particular. This also negatively affects the image of the country to potential international partners.

Administrative, political and economic deterioration negatively affects both domestic and foreign savings and investment. As a result, talented people and investors leave the country for better conditions elsewhere.

The slowdown in the world economy and corporate strategies in favour of relocating production to lower cost locations lead to a rise in protectionism. Anti-dumping measures other than tariff barriers and widespread use of investment incentives by developing countries compound the situation in Lesotho by creating obstacles to exports from Lesotho and making it difficult to attract export oriented FDI. At the same time the Official Development Assistance drops steadily because of the worsening national capacity to implement donor funded projects.

This is *Lesupi* - the worst-case scenario, with a long-term average growth rate of less than 1 percent.

#### 4.3 The Preferred Scenario

The *Melupe* scenario, which is the best-case scenario, portrays the desired future for the Kingdom. It is therefore, the preferred scenario for the country. Thus, it is the one towards which grand and thematic strategies will be geared. The realisation of this scenario will result in stable democracy, a united and prosperous nation at peace with itself and its neighbours. Furthermore, it will bring about a healthy and well-developed human resource base with strong economy, well-managed environment and well-established foundations for technological advancement.

#### 4.4 Conclusion

The unfolding of these scenarios could be influenced by actions and inactions of government departments regarding the implementation of policies and programmes in important areas such as HIV and AIDS, education, environmental management, and relations with the Republic of South Africa. These scenarios are not predictions. They are possible pathways that the future development of Lesotho could take, depending on some key assumptions.

What the scenarios show is that a critical uncertainty in Lesotho's future development relates to the capacity of government to provide sustained support to the country's

dev	elopmen	t process,	and the	country'	s ability	to	stimulate	local	investment,	attract	foreign	direct
inve	stment, a	nd improve	access	to export	t market	S.						



# CHAPTER 5 NATIONAL DEVELOPMENT STRATEGY

#### 5.1 Introduction

This Chapter presents the elements of Lesotho's national development strategy to achieve the Lesotho Vision 2020. The strategy is designed at two different levels. First is an overall or grand strategy, which provides the basis and focus for thematic strategies. Second are thematic strategies, which are mapped out to address the components of the Vision statement. These two levels should guide Government Ministries, civil society and the private sector to formulate their respective strategies.

The strategies are based on the assessment of the current development environment as well as assumptions about its possible evolution in future. These strategies have been developed as responses to key strategic challenges. Strategic issues are fundamental concerns that must be addressed in order to achieve development goals. These challenges are derived from the internal or external environment. Strategic issues have significant impact on long-term development performance, have major multi-sector implications and require the attention of top decision makers. Based on the Lesotho Vision, as well as the analysis of strengths, weaknesses, opportunities and threats, a number of strategic issues have been identified. The issues cut across economic sectors and thematic areas and require attention at the highest levels of national leadership.

## 5.1.1 The Grand Strategy

The grand strategy for Lesotho Vision 2020 addresses the following three key issues:

- How to ensure sustained political commitment and support to the implementation of Lesotho Vision 2020;
- How to ensure sustained high levels of investment; and
- How to strengthen development management capacity.

As an over-arching national development strategy, the grand strategy deals with political commitment and sustained support for the Lesotho Vision 2020. This is crucial especially given the volatile political history of Lesotho, which was characterized by discontinuity and

violence as governments changed one from the other. Establishment of appropriate structures will therefore, enhance the implementation of the Lesotho Vision 2020.

Sustaining high levels of local and foreign direct investment and strengthening business linkages is imperative. Thus, there are comprehensive strategies on how the country should accomplish this important goal. Finally, the grand strategy deals comprehensively with development management, which is one of the key challenges that the country faces.

Matrices 5.1 to 5.3 present the strategic actions, monitoring indicators, means of verification, and the main actors.

## 5.1.2 Thematic Strategies

In the context of the grand strategy there is another level of strategies including thematic strategies. Matrices 5.4 to 5.10 present key strategic actions, monitoring indicators, means of verification and main actors for achieving the key Vision Aspirations. These strategies are built around clusters of Vision Statement components as follows; stable democracy which entails governance, gender and media; a united nation which includes culture; nation at peace with itself and its neighbours; healthy and well developed human resource base which includes health, HIV and AIDS and education; strong economy and prosperous nation; well managed environment; and well established technology. These strategies address strategic challenges for the country and will be implemented by different sectors of the economy as indicated in each of the matrices below.

Matrix 5.1
Sustaining Political Commitment and Support

Strategic Action	Indicator	Means of Verification	Main Actors
Resuscitate National Planning Board as an advisory body and inject Vision philosophy into it.	<ul> <li>Amended constitution</li> <li>Regular meetings of the National Planning Board where Vision concerns is one of Agenda items.</li> <li>Implementation rate of decisions</li> </ul>	- Minutes of the meetings of the National Planning Board.	Ministry of Finance and Development Planning, and Parliament.
Establish Vision Supporting Institutions     Parliamentary Working Committee on Planning     Think Tank on Public Policy, Research & Analysis	Existence of the institutions     Implementation rate of decisions	<ul><li>Hansard Records,</li><li>Records of meetings</li><li>Publications</li><li>Seminars</li><li>Conferences</li></ul>	Research Institutions, Civil Society and National Assembly,
Integrate Vision concerns into plans and programmes of all Government Ministries and civil society organisations	- Guidelines on integration of Lesotho National Vision concerns into sector plans and programmes	Written directive from the Prime Minister's Office.  Development plan implementation reports  Programme appraisal reports	Ministry of Finance & Development Planning, Prime Minister's Office, Private Sector and Civil Society

Matrix 5.2
Sustaining High Levels of Investment

Strategic Action	Indicator	Means of Verification	Main Actors
Improve and sustain good investment climate	Existence & Implementation rate of the Investment Policy     Availability of accessible finance scheme     Improved Border controls	Investment policy document     National Income and Product     Accounts	Ministry of Industry and Trade, Cooperatives and Marketing,     Private sector, LNDC, LTDC, Cooperatives
Strengthen the Private - Public interface in building local capacity to sustain Investment.	<ul> <li>Existence &amp; success rate of Public-Private Partnership Agreements</li> <li>% of locals in managerial positions</li> <li>% of locally owned factories as % of total investors</li> </ul>	- Investment Records	Ministry of Trade and Industry, Cooperatives and Marketing, Private sector, LNDC, LTDC, Cooperatives and Parastatals
Explore and exploit investment potential in the services and natural resources sectors to maximise comparative advantage.	Research studies     Increased local product range for exports	Research and Investment reports     Foreign trade reports	Research institutions and Ministry of Natural Resources
Strengthen provision of infrastructure to the industrial estates.	The number of fully serviced industrial estates     Existence of transportation network facilities	- Infrastructural network maps	Ministry of Natural Resources, Parastatals, City Authorities and Ministry of Works
Promote competitiveness and diversification in the industry sector	Range and Quality of Products     Number of Market Outlets	Trade Statistics and Reports     Labour Statistics	Ministry of Industry and Trade, Cooperatives and Marketing, Private sector, LNDC, LTDC, Cooperatives
Strengthen investment and trade promotion capacity of the country	- Existence of fully resourced and well co-ordinated investment promotion institutions	- Trade Promotion and Investment Reports	Ministry of Industry and Trade, Cooperatives and Marketing, Private sector, LNDC, LTDC, Cooperatives
Develop and promote backward and forward linkages between foreign owned firms and local businesses	- Rate of success of the linked firms	- Trade and investment reports	Ministry of Industry and Trade, Cooperatives and Marketing, Private sector, LNDC, LTDC, Cooperatives
Diversify export markets	Number and adequacy of alternative export markets and agreements the same as AGOA	- Copies of new agreements; periodic reports. On export markets and agreements	Ministry of Industry and Trade, Cooperatives and Marketing, Private sector, LNDC, LTDC, Cooperatives

Matrix 5.3

Strengthening Development Management Capacity

Strategic Action	Indicators	Means of Verification	Main Actors
Improve public financial management and Budgeting	<ul> <li>Existence &amp; implementation rate of prudent fiscal strategies</li> <li>Sustainable fiscal policy</li> <li>Integrated financial management systems</li> </ul>	<ul><li>Fiscal reports</li><li>Auditors reports</li><li>Debt reports</li></ul>	Ministry of Finance and Development Planning Public Accounts Committee (PAC) Auditor General
Develop an integrated planning and budgeting system	<ul> <li>Existence of the planning system and implementation of the plan</li> <li>Operational efficiency of public expenditure</li> </ul>	Computer packages     National reports	Ministry of Finance and Development Planning, Private Sector and Civil Society
3. Improve human resource management	<ul> <li>Existence &amp; implementation rate of Training &amp; Development Policy</li> <li>Success rate of Performance Management System</li> <li>Existence of Integrated Human Resource Management Systems</li> </ul>	Training & Development Policy     Performance Management Reports	Prime Minister's office and Ministry of Public Service
Decentralize service delivery and empower communities	<ul> <li>Implementation of the Local Government Act</li> <li>No. of trained &amp; empowered local councils and communities</li> <li>Success rate of community development projects</li> </ul>	- Local Government Elections records	Ministry of Local Government, Local governance structures and Donor community
Strengthen & promote private sector and civil society participation in managing development	<ul> <li>Existence &amp; implementation of a civil society development Plan</li> <li>Existence of enabling environment for the private sector to participate effectively</li> </ul>	- Private Sector & Civil Society records	Government Secretary, Civil Society and Private Sector
6. Strengthen aid coordination	<ul> <li>Aid Policy</li> <li>Adherence to Aid policy guidelines in implementation of programmes</li> </ul>	Policy records/Document     Programme implementation and reports	Ministry of Finance and Development Planning Donor community
Strengthen research capacity in the country	<ul><li>Quality and number of research reports and publications,</li><li>% of national budget allocation to research</li></ul>	- Published and unpublished reports	Universities, Research institutions, consultants
Eradicate corruption and nepotism in all sectors of the economy	- Rate of decrease in corrupt practices	- Court records - Audit reports	Anti Corruption Unit Ministry of Justice, Human Rights and rehabilitation and of law and constitutional affairs General public

## Matrix 5.4

## A Stable Democracy

Strategic Actions	Indicators	Means of Verification	Main Actors
Develop and promote visionary leadership in the country	<ul> <li>Quality of leaders decisions</li> <li>Number of disciplinary cases on breach of conduct</li> <li>Number of leaders effectively trained</li> </ul>	Code of conduct  Evaluation & Monitoring reports	Civil society, Prime Minister's office and College of chiefs
Promote popular participation in national affairs	<ul><li>% of populace trained in civic education</li><li>% of voters participating in elections</li></ul>	Training & Evaluation Reports, Campaigns etc	Civil Society, Local Governance Structures, Ministry of Local Government
3. Strengthen governance institutions	Institutional autonomy     Success rate of governance institutions recommendations	Budgets, Plans and Records	Prime Minister's Office
Promote speedy dispensation of Justice	Effective Coordination mechanisms between Police and Courts of law     Rate at which cases are decided	Procedures and guidelines Records and reports	Law associations, DDPR, Ministry of Justice, Human Rights & Rehabilitation & of Law & Constitutional Affairs, Local Authorities.
5. Increase media coverage	- % media coverage in country	Survey reports	Media institutions and organisations, and Ministry of Communications, Science & Technology
Promote media professionalism and independence	Existence & implementation of Media Policy     Increased number of trained media practitioners	Policy documents  Training Records and code of conduct	Media Institutions and organisations, and Ministry of Communications, Science & Technology
7. Consolidate media laws	- Existence of document on consolidated media laws	Hansard reports	Media institutions and organisations, and Ministry of Communications, Science & Technology, Parliament
8. Promote gender equality.	- % of women in Parliament, in decision making positions -	- Revised laws - Labour statistics	Civil Society, Ministry of Gender, Youth, Sports & Recreation and Parliament
Implement the Local Government Act	- Functional Local Government Structures	- Policy Document	Ministry of Local Government, Chiefs, Civil Society and CBOs
10. Strengthen chieftainship as an institution of governance	<ul><li>Effective local governance structures</li><li>Opinion Survey results</li></ul>	- Records	Ministry of Local Government and College of Chiefs

## Matrix 5.5

## **A United Nation**

Strategic Actions	Indicators	Means of Verification	Main Actors
Promote national unity	Widespread use of symbols, language & traditions	Newspaper reports and research findings	Civil Society, Ministry of Tourism, Environment and Culture, Tourism Authority and Media houses
2. Promote Cultural Practices	<ul> <li>Existence and implementation of cultural policy</li> <li>Cultural curriculum development</li> <li>National Cultural Days</li> <li>Establishment of Cultural Research Unit</li> <li>Regulated initiation institutions</li> </ul>	<ul> <li>Records</li> <li>Preserved cultural centres</li> <li>Policy document</li> <li>Curriculum</li> <li>Legal notice declaring cultural day</li> <li>Research Unit and implementation records</li> </ul>	Ministry of Tourism, Environment & Culture, Tourism Authority and civil society and Ministry of Education & Training, cultural interest groups
Preserve and protect cultural heritage resources	- Number of preserved and protected cultural heritage resources	- Records - Preserved cultural centres	- Ministry of Tourism, Environment & Culture, Tourism Authority and civil society and Ministry of Education & Training, cultural interest groups
4. Instil the spirit of patriotism	<ul> <li>Number of skilled people staying in the country</li> <li>Rate of contribution of exported labour force</li> <li>Rate of participation in cultural practices</li> </ul>	Reports on skilled labour retained     Amounts and flow of remittances from skilled labour	Private sector, Ministry of Labour and Employment, Recruiting agencies     Youth associations
Research and document cultural practices	- Number and quality of research reports	Research reports     Documents on cultural practices	- Research institutions, Private sector, Schools, Ministry of Tourism, Environment and Culture, Ministry of Education and Training.
6. Promote a healthy family life	- decreasing divorce rate  - decreasing domestic violence	- records	- Churches, Ministry of Justice, Human Rights and rehabilitation and of law and constitutional affairs

Matrix 5.6

A Nation at Peace With Itself and Its Neighbours

Strategic action	Indicator	Means of Verification	Main Actors
Transform & professionalize security establishments	- Success rate of professional security establishments programmes	- Coordination systems and procedures	Ministry of Defence & National Security
	- Coordinated functions among security establishments		Ministry of Home Affairs and Public Safety
			Ministry of Justice, Human Rights and rehabilitation and of law and constitutional affairs
2. Combat crime.	<ul><li> Crime statistics.</li><li> Prisons statistics</li></ul>	- Statistical reports	Ministry of Home Affairs and Public Safety and Civil Society, Ministry of Justice, Human Rights and Rehabilitation and of Law and Constitutional affairs
Create conflict resolution mechanisms at all levels of society	<ul><li>Success rate of conflict resolution structures</li><li>Existence of conflict mgt structures at all levels</li></ul>	- Reports and records - Structures	Prime Minister's office, Civil Society
Promote frameworks of strategic collaboration with neighbouring countries	<ul> <li>Number of bi-lateral agreements and implementation rate of agreements</li> <li>No. of conventions ratified by the country</li> </ul>	- Agreements	Ministry of Foreign Affairs

# A Healthy and Well Developed Human Resource Base

Matrix 5.7

Strategic Actions	Indicators	Means of Verification	Main Actors
Formulate and implement Health and Social welfare policy framework	- Implementation rate of Health Policy	Health policy document and reports	Ministry of Health and Social Welfare, CHAL, Private Health institutions
2. Improve access to health care	Distance travelled to health care facilities     Financial access	- Records	Ministry of Health and Social Welfare, Private Health Institutions
Improve access and control of pharmaceuticals	<ul> <li>% of local demand met by local supply</li> <li>Existence and implementation rate of drug policy</li> </ul>	- Drug policy documents	Ministry of Health and Social Welfare, Pharmaceutical Co-operations
Increase provision of clean, safe drinking water and proper sanitation	Increased coverage of population with access to drinking water and sanitation	- Public Health Records	WASA and Ministry of Natural Resources
5. Reduce HIV and AIDS infection rate	<ul><li>Distribution and usage of contraceptives</li><li>% of population infected</li><li>Awareness campaigns</li></ul>	Awareness campaign reports	HIV and AIDS authority, Community Support Groups, CHAL, all Government Ministries and Private Sector
6. Create an HIV competent society	<ul> <li>Rate of change in peoples of change behavioural patterns</li> <li>People equipped with life skills</li> <li>Rate of decline in HIV and Aids infections</li> </ul>	- Reports on HIV and AIDS situation	Families, CHAL, Community Support Groups, Private Sector, All Government Ministries
Provide prevention, care & support for the HIV affected and infected	<ul><li>Social protection policy</li><li>Safety nets for AIDS Orphans</li></ul>	Policy document and legislation	Ministry of Health and Social Welfare, Civil Society and Donor Community, Families
8. Promote Lesotho as a service industry	<ul> <li>% of skilled human resource exported to other countries</li> <li>Contribution of income from abroad to GNP</li> <li>Strategic plan for retention of rare skills</li> </ul>	- National income & product accounts	Prime Minister's Office and Ministry of Employment & Labour, Educational and Research institutions
9. Develop National Manpower Development Plan	- Existence and effectiveness of the Manpower Plan	- Manpower reports	Ministry of Finance & Development Planning, Institutions of higher learning, Private sector and Parastatals

10. Improve Education Infrastructural Facilities	- Number of educational institutions with improved facilities	- Education Records	Church and Private schools, Teachers' Associations and Ministry of Education,
11. Develop educational curriculum responsive to the country's needs	- % of graduates absorbed into the labour market	- Employment records	Ministry of Education & Training, churches, and private, teachers' association
12. Improve Institutional management and human resource capacity	<ul><li>Pass rates</li><li>% of qualified teachers</li></ul>	<ul><li>Examination results</li><li>Statistical Education records</li></ul>	Ministry of Education & training, Donor Community, Teachers' associations
13. Promote Special education Programmes for people with disabilities	- Number of equipped special training institutions	- Special Education Records	People with Disabilities and Ministry of Education & Training
14. Strengthen vocational training	<ul><li>Number of students in vocational training institutions</li><li>% Budget allocation to vocational training</li></ul>	<ul><li>Institutions' Reports</li><li>Financial reports</li></ul>	Churches, Ministry of Education and Training, Private sector
15. Build sporting facilities and strengthen capabilities	<ul><li>number and quality of facilities</li><li>Number of youth engaging in sports</li><li>number of medals</li></ul>	- reports and sports records	LSRC, LNOC, Ministry of Gender, Youth, Sports and Recreation, Youth associations

Matrix 5.8

Strong Economy and Prosperous Nation

Strategic Action	Indicators	Means of Verification	Main Actors
Improve macroeconomic management     Strengthen expenditure control and financial management     Strengthen fiscal transparency	<ul> <li>Existence &amp; implementation of financial systems and procedures</li> <li>GDP growth rate</li> <li>Debt service ratio</li> <li>Fiscal balance</li> </ul>	<ul> <li>Software packages and records</li> <li>Balance of payments accounts</li> <li>Audit reports</li> <li>National accounts</li> </ul>	Ministry of Finance & Development Planning
Strengthen Financial Services sector     Encourage more competition in the sector through more banking, capital markets and insurance	<ul><li>No. of financial intermediaries</li><li>Investment/ Savings ratio</li></ul>	- Central Bank records	Ministry of Finance & Development Planning, Central Bank of Lesotho, Private Sector and LRA
Develop Tourism Potential     Implement Tourism Policy	Success rate of implementing Tourism Policy     Contribution of tourism to GDP	Plans, records and reports	Ministry of Tourism, Environment & Culture, Private sector and LTDC
4. Strengthen and Promote SMME  - Develop and implement the SMME Policy  - Establish clear roles between the public and private sector in the provision of essential business development services: Financial resources, business management and technical skills and infrastructure.	<ul> <li>SMME Policy</li> <li>Improved licensing procedure</li> <li>Responsive export fincance schemes</li> <li>Increased number of professional service providers</li> </ul>	Policy document	Ministry of Ministry of Industry and Trade, Cooperatives and Marketing, Business associations, Central Bank of Lesotho, Cooperatives
5. Implement PRSP and the MDG Programmes	- Change in poverty levels - Poverty index	Plans, records, reports	Ministry of Finance and Development Planning and Donor community
Create growth and employment opportunities in the labor-intensive sectors.     Strengthen employment opportunities in the manufacturing sector     Develop tourism potential taking into account the available natural resources, scenic beauty of the country, handicrafts development, pony trekking and forge	<ul> <li>Employment rate</li> <li>% of total labour force employed in the manufacturing sector</li> <li>Number of tourism enterprises owned and run by Basotho people</li> <li>% of total labour force employed in the tourism sector</li> </ul>	- Labour survey documents	Ministry of Finance and Development Planning, Ministry of Employment & Labour and Private sector

<ul> <li>7. Improve food security and overcome hunger</li> <li>Combat wide-spread stock theft.</li> <li>Diversification into cash crops</li> <li>Revision of land tenure system</li> <li>Advance irrigation methods and effective extension service.</li> </ul>	<ul> <li>ha. of cultivated land and yield</li> <li>% of land under irrigation</li> <li>Crop diversification</li> <li>Number of farmers per extension worker</li> </ul>	- Agricultural statistics	Ministry of Agriculture & Food Security and Private sector
Reallocate government resources to expenditure with higher impact on poverty.     Develop and implement a Policy on social security and safety nets     Develop safety nets for Aids orphans	<ul><li>Increased budget Social Welfare budget</li><li>Social Welfare and Protection Policy</li></ul>		Ministry of Finance and Development Planning  Ministry of Health and Social Welfare

Matrix 5.9

## **Well Managed Environment**

Strategic Actions	Indicators	Means of Verification	Main Actors
Strengthen coordination of institutions responsible for natural resource management	No. of operational Environmental Units	Environment Reports	Ministry of Tourism, Environment and Culture and LTDC
2. Protect air, soil and water quality	Existence & implementation of legislation and Policy	Policy documents	Ministry of Tourism, Environment and Culture and LTDC
Develop and implement proper land management systems	Successful implementation of land use plans	Environment Monitoring reports	Ministry of Forestry & Land Reclamation, Ministry of Local Government and CBOs
Preserve and conserve biodiversity and heritage	Number of protected areas	Environment reports	Ministry of Tourism, Environment and Culture and LTDC
Improve institutional capacity to implement the Environment Act	Well resourced ministerial environmental units	Environment programmes and reports	Ministry of Tourism, Environment and Culture, the Private Sector and LTDC
Reduce over-harvesting of natural resources	Existence & implementation of regulations	Policy documents	Ministry of Natural Resources and CBOs
Improve effluent and solid waste management systems	Effective Effluent management standards Waste recovery rates	Waste management Standards	City Authorities, WASA and Ministry of Tourism, Environment and Culture and CBOs
8. Promote use of renewable energy resources.	Diverse and environmentally friendly energy resources	Environmental reports	Ministry of Natural Resources and Private Sector

## Matrix 5.10

## **Well Established Technology**

Strategic Actions	Indicator	Means of Verification	Main Actors
Strengthen education and awareness on Science and Technology	<ul><li>S &amp; T expenditure as % of GNP</li><li>Implementation rate of S &amp; T policy</li></ul>	<ul><li>National accounts</li><li>Monitoring and evaluation reports</li></ul>	Educational institutions and Ministry of Communications, Science and Technology,
Strengthen research and development	<ul><li>Existence of S &amp; T information centre</li><li>No. of publications and patents</li></ul>	<ul><li>S &amp; T reports</li><li>Publications and patents</li></ul>	Tertiary and research institutions, Appropriate Technology Section and Ministry of Communications, Science and Technology,
Commercialize indigenous technological knowledge and practices	- Implementation rate of S & T policy	- Monitoring and evaluation reports	Ministry of Communications, Science & Technology and Appropriate Technology Section, Private Sector
Develop and implement ICT policy	- ICT policy implementation and evaluation	ICT policy document     Monitoring and evaluation reports	Ministry of Communications, Science and Technology, Tertiary and research institutions and Private sector



#### **CHAPTER 6**

# IMPLEMENTATION, MONITORING AND EVALUATION

#### 6.1 Introduction

The Lesotho Vision 2020 process has enjoyed commitment and support at the highest levels. It has also involved the widest possible consultation with stakeholders. The output of this challenging national exercise is the Strategy that has been presented in Chapter 5. Thus far, the process has led to the formulation of a strategic framework which is an indicative plan to guide development policies, programmes and activities.

In Chapter 5 the key strategic actions needed to operationalize Lesotho Vision are identified. Chapter 6 identifies indicators and spells out goals for the Vision components. It also identifies the activities needed to kick-start the implementation process for ensuring effective monitoring and evaluation. To this end therefore, the Chapter explains as cornerstones of the Vision, implementation, monitoring and evaluation, indicators and implementation plan and concludes with a statement about Vision – supporting institutions to be in place.

## 6.2 Implementation, Monitoring and Evaluation

#### 6.2.1 Implementation

Matrices 5.1 to 5.10 presented the strategic actions, monitoring indicators, means of verification for addressing the issues concerning the grand strategy as well as the thematic issues that underpin the Lesotho Vision. In implementing these strategic actions, the following challenges must be recognized:

- Integrating plans for different time horizons: Lesotho Vision 2020, rolling plans, PRSP and annual budget statements;
- Sustaining the strategic thinking process in order to ensure that emerging strategic issues are identified and anticipated;
- Monitoring and evaluating the Lesotho Vision 2020 and other plans.

#### a) Integrating Different Plans

As noted in Chapter 1, the Lesotho Vision 2020 is a strategic framework that focuses on the key issues that need to be addressed to ensure that Lesotho realises its development vision. The successful implementation of Lesotho Vision depends, to a large extent, on the effective implementation of operational plans as contained in three-year rolling plans, PRSP, ministerial and departmental plans, and the annual budget statements. This can be ensured if appropriate institutional arrangements are made to enhance operational effectiveness of government ministries and departments in reflecting Lesotho Vision 2020 strategies in their development programmes, and in creating the enabling environment for private sector growth and development. In turn this requires that:

- Those assigned the responsibility to address strategic issues and actions under Lesotho Vision should elaborate detailed plans and allocate responsibilities to collaborating ministries, departments and agencies;
- Ministries, departments and agencies reflect their operational plans, as appropriate in their three-year rolling plans, PRSP implementation, policy papers, legislation and annual budget submissions;
- The civil society organisations and the private sector reflect as appropriate, vision issues in their respective plans.

## b) Sustaining Strategic Thinking

The preparation of the Lesotho Vision 2020 involved a survey and analysis of the internal and external environment affecting the future socioeconomic development of Lesotho. The purpose was to identify strengths, weaknesses, opportunities and threats that should inform strategic choices. Clearly, both the internal and external environment are subject to change, resulting in the emergence of new opportunities and challenges. It is thus necessary to sustain the process of collating and analyzing strategic information and using it to update the national development strategy. A key element of the strategic thinking process is the establishment and maintenance of appropriate information system, including the strengthening of institutional capacities to enhance strategic foresight in development management.

#### 6.2.2 Monitoring and Evaluation

It is often pointed out that Lesotho prepares good plans but is unable to implement them effectively. One of the characteristics of a well-formulated plan is that the plan must address issues relating to effective implementation, monitoring and evaluation. The process has to be planned, implemented, monitored and evaluated.

The monitoring and evaluation plan for the Lesotho Vision 2020 should include the following activities:

- Preparing detailed implementation plans for identified strategic actions;
- Establishing consensus on indicators for monitoring and evaluation;
- Agreeing on how to generate and utilize information for monitoring and evaluation;
- Specifying reporting requirements;
- Establishing monitoring and evaluation responsibilities; and
- Providing adequate budget for monitoring and evaluation.

## 6.2.3 Indicators and Implementation Plan

Matrix 6.1 presents the main components of the Lesotho Vision 2020 statement, relevant indicators for monitoring, baseline information, as well as goals for three time periods to 2020. The matrix identifies indicators for monitoring progress in the achievement of Vision goals.

A wide range of monitoring indicators has been identified for monitoring progress in various aspects of the Lesotho Vision. Clearly, it will not be possible or feasible to establish baseline data and regularly monitor all the indicators. The challenge during implementation will be to review the list of proposed indicators and to select the minimum requirements in the light of information available.

Matrix 6.1

Monitoring Components of the Lesotho Vision 2020

Vision	Indicator	Baseline	Goal		
Component			2010	2016	2020
Stable democracy	<ul> <li>Civil liberties</li> <li>Political Rights</li> <li>Press Freedom</li> <li>Rule of Law</li> <li>Voter Turnout</li> <li>Corruption Perceptions Index</li> <li>Polity Score</li> </ul>	4 (2000) 4 (2000) 52 (2000) -0.19 (2000-2001) 74% (1998) - Not Available - (2000)	3.5 3.5 47 0 79 4	3 3 42 0.05 84 7	2.5 2.5 37 1 89 10 6
United nation	Opinion survey     Stable and professional Security Establishments	- (2002) 45% (2003)	1 58.5%	2 64.5%	+2 70%
Nation at peace with itself and its neighbours	<ul> <li>Political Court cases</li> <li>Political Prisoners</li> <li>Crime statistics</li> <li>Cross border crime</li> <li>Implementation rate and adherence to bilateral and multilateral agreements</li> </ul>	17,836 (2003)	14,268.8	11,593	8,918
Healthy and well	- Human Development Index (HDI)	0.54 (2000)	0.58	0.63	0.68
developed human resource base	<ul> <li>Life expectancy at birth</li> <li>Adult literacy rate</li> <li>% of population covered by social protection</li> </ul>	45.7 years (2000) 83.4% (2000) -	40 87.6	60.9 91.7	68.4 95.9
	<ul> <li>Popn. Using improved water sources</li> </ul>	91% (2000)	93%	97%	100%
	- Popn. with sanitation services	92% (2000)	93%	97%	100%
	<ul> <li>Popn with access to essential drugs</li> <li>People with HIV AIDS</li> <li>Health expenditure as % of GNP</li> <li>Education expenditure as % of GDP</li> </ul>	80-94% (1999) 31% (2000) 8.4% (1997) 20% (1990)	96% 24.5% 30%	98% 17% 30%	100% 9.5% 30%

Ctrons	CDD succeeds note	2.00/ (2002)	2.50/	70/	70/
Strong economy	- GDP growth rate	3.8% (2003)	3.5%	7%	7%
and	- Consumer Price Index change	6.1% (2000)	6.1 %	6.0%	5.9%
Proeporous		3.35 (2003)	3.22	3.10	2.97
nation	- Share of Secondary sector to GDP	40% (2000/01)	55%	70%	70%
	- Competitiveness Index	Not available			
	- Investment as % of GNP	33% (excl. LHWP) (1997-2001)	39%	45%	51%
	- Saving as % of GNP	20% (1997-2001)	35%	45%	55%
	- Exports of goods and services as % of GDP	28% (2000)	35%	41%	48%
	- Imports of goods and services as % of GDP	88% (2000)	82%	75%	70%
	- Current Account Balance as % of GNI	7.9% (2003)			
	- Fiscal Balance as % of GDP	4.2% (2003)			
	- Level of external reserves	6 months	6 months	6 months	6 months
	- Debt Service Ratio				
	- Human Poverty Index	26(2000)	21	15	10
	- % of popn. below poverty line	58%(2000)	44%	29%	15%
	- Gini Index	57% (1996)	50%	40%	25%
	- Per capita GNI	\$550 (2001)	\$600	\$650	\$700
	- Unemployment rate	31% (1999)	29.5%	18.0%	15.%
	- Household Food Security	10% h/h in commercial intensive livestock production (1997)	25%	30%	35%
	- Irrigation	6000ha small scale irrigation (1997)	10,000ha	15,000ha	20,000ha
Well	- Arable land	9% (1998)			
managed	- Land area under forestation	10,362ha (2000)	13,550ha	16,380ha	21,000ha
environment	- Rangeland	1,437,214ha (2000)	1,580,214ha	1.738,214 ha	1,911,214ha
	- Solid Waste	143,600 tonnes(2001)	150,000 tonnes	153,000 tonnes	158,000 tonnes
	- Waste Recovery rates	63% (1997)	70%	75%	79%
Technology well	- National Expenditure on S&T as % of GNP	Less than 1% (2002)	1.5%	2.0%	2.5%
established	- % of Science & Maths students in tertiary institutions	13% (2000)	16.25	19.50	26%
	- % of government budget allocated to S&T				
	- Telephone & cellular lines per 1000 people	10(2000)	20	35	50

#### 6.3 Vision Supporting Institutions

The following are the proposed institutions that will support implementation of short, medium and long-term plans including the Lesotho Vision. They include the Parliamentary Committee on Planning, National Planning Board and a privately-run Think Tank. The rationale for these institutions is effective implementation of the Vision. Furthermore, it is important and operationally useful that there are institutions mandated to check and balance one another in the monitoring of implementation of this important long-term national plan. This is in line with the fact that Lesotho's greatest challenge in development management is plan-implementation and evaluation. The following organisational structure indicates the envisaged way in which these institutions will relate with each other.

## Vision Supporting Institutions' Relational Structure



## 6.3.1 Functions of Vision Supporting Institutions

### a) Parliamentary Committee on Planning

It is necessary for Parliament to be able to keep track of the implementation of the National Vision. In the light of this it is imperative that there be a committee within Parliament that will be dealing specifically with issues pertaining to planning that is in turn geared towards the achievement of the Vision.

#### b) National Planning Board

As per the constitution, the National Planning Board will guide, monitor and implement short, medium and long-term plans. These plans will include PRSP, MDGs and Lesotho Vision 2020.

#### c) Think Tank

There should be a privately managed and independent think-tank that will scan the environment on a continuous basis through research, seminars and conferences, and advise government accordingly on, among others, policy issues and national development strategy implementation. One of the main advantages of a body of this nature is that short and medium-term plans will be closely monitored in relation to their relevance and contribution to the achievement of the Lesotho Vision 2020.

# 6.3.2 Implementation, Monitoring and Evaluation Plan

Matrix 6.2 identifies the key activities for kick-starting the process of implementation, monitoring and evaluation of Lesotho Vision 2020.

Implementation, Monitoring and Evaluation Plan

Matrix 6.2

No.	Activity	Start Date	Finish Date	Output	Main Actors
1.	Set up Implementation Task Force	Last quarter 2004	End of December 2004	Structure working procedures manuals	Government Secretary's office,
2.	Prepare action plan for attracting FDI	Last quarter 2004	End of December 2004	Action plan	PS Trade, BEDCO, LNDC and Private Sector
3.	Prepare action plan for sustaining political commitment and support	Last quarter 2004	End of December 2004	Plan	Prime Minister's Office
4.	Prepare action plan for strengthening development management	Last quarter 2004	End of December 2004	Plan	Ministries of Public Service, Finance & Development Planning and Local Government
5.	Establish Vision supporting institutions	3 <sup>rd</sup> quarter 2004	Second quarter 2005	Structure working procedures manuals	Prime Minister's Office and Parliament
6.	Prepare terms of reference for Vision supporting institutions	Last quarter 2004	End of December 2004	Terms of reference	Government Secretary and Private Sector
7.	Establish baseline data for monitoring Lesotho Vision 2020	Last quarter 2004	End of December 2004	Database	Ministry of Finance and Planning
8.	Finalise implementation, monitoring and evaluation plan for Lesotho Vision	Last quarter 2004	End of December 2004	Monitoring and evaluation plan	Government Secretary
9.	Revise Planning Officers Manual to incorporate chapter on strategic planning and Lesotho Vision 2020.	Last quarter 2004	End of December 2004	Revised Planning Officers Manual	Ministry of Finance and Development Planning
10	Prepare Action Plan to sell the Vision to the nation	Last quarter 2004	End of December 2004	Plan	Government Secretary

## Annex 1

## **Lesotho Vision 2020: Critical Success Factors**

Lesotno Vision 2020: Critical Success Factors					
Stable Democracy	Nation at Peace With Itself and Its Neighbours				
Governance	Peace				
Constitutionalism	Crime Prevention				
Local governance	Social cohesion				
Political commitment to the Vision process	Safety and Security				
Political stability	Conflict Management				
Parliament	Sommor management				
Judiciary	Healthy and Well Developed Human Resource Base				
Democratic institutions	Education				
Civil society	Budget allocation				
Development Management	Relevant curriculum development				
Strategic management	Accessibility to education				
Policy management	Strengthening of lifelong, TVET, special, tertiary education				
Public expenditure management	Infrastructure development				
Public sector productivity	Institutional management				
Programme management	Human resource capacity				
Aid management					
Information systems	Health				
Institutional leadership	Access to health care services				
Media	Control and management of pharmaceuticals				
Freedom of expression	Human resource availability and development				
Media freedom	Infrastructure development				
Professionalism within the media	Social welfare management				
Gender	Institutional management capacity				
Equality of opportunities					
Law reform	HIV and AIDS				
Gender and development policy	Control of infection				
Basotho cultural practices	Care and support for infected and affected				
	Strategies and policies development				
United Nation	Funding for HIV and AIDS				
Culture	Community education on HIV and AIDS				
Preservation and promotion					
Institutional strengthening					
National values					
Language empowerment					
Strong Economy and Prosperous Nation	Well Managed Environment				
Economy	Environment				
Macroeconomic management	Environmental management				
Economic growth	Land management and desertification				
Inflation	Conservation and biodiversity				
Savings	Waste management				
Investment	Conservation of mountain ecosystems				
Government revenue/expenditure	Climate change				
Exports & market access					
Economic infrastructure	Well Established Technology				
Public debt	Improved management capacity				
SMME	S&T human resource development				
Employment	Technological diffusion				
Economic structure	Promotion an d commercialization of indigenous				
Prosperity Food counity	knowledge and practices				
Food security	Research and development				
Shelter	Education and awareness on S&T				
Community empowerment					
Access to financial services					
Distribution of income and wealth					
Poverty reduction & Population dynamics					

#### Annex 2

#### **Scenario Drivers and Assumptions**

#### 1. Focal Issue

Future development of Lesotho

#### 2. Driving Forces

- Sustained political commitment and support for Lesotho National Vision initiative
- Political stability
- Development management capacity, including public sector management, and creating enabling environment for private sector development
- Domestic saving
- Human resource development and utilisation
- Investment
- Export markets
- Natural resources management

#### 3. Factors behind driving forces

- Motivation and commitment of public servants
- Transparency and accountability in public management
- Planning systems, project planning, implementation capacity
- Leadership and supervision at key levels of administration
- Financial intermediation, savings instruments, bank supervision
- Tilianciai intermediation, savings instruments, bank superv
- Tax systems and structure
- Labour relations
- Balance between tertiary education and primary education
- Demographic factors, migration
- HIV and AIDS
- Economic and social infrastructure
- Productivity and competitiveness
- Effective Public Expenditure Management

#### 4. Key Trends

- Declining foreign aid
- Worsening income distribution and inequality
- Declining SACU revenue, miners' remittances
- HIV and AIDS

#### 5. Key Uncertainties

- Public sector management
- Future trends in investment
- Political stability

#### 6. Alternative Scenarios

**Scenario 1:** Political stability, effective public sector management and favourable trends in investment (the *Melupe* Scenario).

**Scenario 2:** Political stability, effective public sector management and declining investment (the *Khomo ha li kae* batho re ba ngata Scenario).

**Scenario 3:** Political stability, favourable trends in investment and weak public sector management (the *Metse e metle* liotloana Scenario).

**Scenario 4:** Political instability, weak public sector management, declining investment (the *Lesupi* Scenario).

# Annex 3

# **Summary of SWOT Analysis**

Major Strengths	Major Weaknesses	Major Opportunities	Major Threats	
- Government Commitment to Development (Lesotho National Vision & PRSP Process) - Constitutionalism - Cultural homogeneity - Electoral System (Proportional representation) - Political Stability - High Adult literacy rate	<ul> <li>Poor Strategic &amp; Operational management</li> <li>Weak institutional leadership</li> <li>Inmjeffective civil society and private sector</li> <li>Slow decentralization and local governance</li> <li>Poor development management information system</li> <li>Weak expenditure and financial management</li> <li>Inadequate S&amp;T research and development</li> <li>Irrelevant curriculum development</li> <li>Poor management of educational institutions</li> <li>Ineffective management of health institutions</li> <li>Underdeveloped SMME sector</li> <li>High unemployment rates</li> <li>Food insecurity</li> <li>Increasing HIV and AIDS infection rates</li> </ul>	<ul> <li>FDI flows</li> <li>Strong Donor support</li> <li>Good Relations with RSA</li> <li>Growing regionalism</li> <li>Bi-lateral relations and multilateral arrangements</li> </ul>	<ul> <li>Brain drain</li> <li>Increasing competition from international markets</li> <li>Donor conditionalities</li> <li>Declining SACU revenues</li> <li>Decline in mine labour remittances</li> <li>Migrant labour retrenchment</li> <li>Ending of global market opportunities</li> </ul>	